# Multiple Unit Auctions and Short Squeezes

Kjell G. Nyborg and Ilya A. Strebulaev

NOTA DI LAVORO 27.2003

#### MARCH 2003 PRIV – Privatisation, Regulation, Antitrust

Kjell G. Nyborg, *London Business School and CEPR* Ilya A. Strebulaev, *London Business School* 

This paper can be downloaded without charge at:

The Fondazione Eni Enrico Mattei Note di Lavoro Series Index: http://www.feem.it/web/activ/\_wp.html

Social Science Research Network Electronic Paper Collection: http://ssrn.com/abstract=XXXXXX

The opinions expressed in this paper do not necessarily reflect the position of Fondazione Eni Enrico Mattei

# **Multiple Unit Auctions and Short Squeezes**

#### Summary

This paper develops a theory of multiple unit auctions with short squeezes in the postauction market. This is especially relevant for financial and commodity markets where players may enter the auction with established forward positions. We study how a potential short squeeze impacts on bidders' strategies and auction performance. Conversely, we also study how the design of the auction affects the incidence of short squeezes. In particular, we model both uniform price and discriminatory price auctions in a true multiple unit setting, where bidders can submit multiple bids for multiple units. Our model is cast in what appears to be a common value framework. However, we show that the possibility of a short squeeze introduces different valuations of the to-beauctioned asset between short and long bidders. Equilibrium bidding strategies depend on pre-auction allocations and the size of the auction. Short squeezes are more likely to happen after discriminatory auctions than after uniform auctions, ceteris paribus. Discriminatory auctions therefore lead to (1) more price distortion; (2) higher revenue for an auctioneer; (3) more volatility in the secondary market. This shows that a central bank or sovereign treasury, say, may face a tradeoff between revenue maximization and market distortions when choosing the design of repo or treasury auctions. The probability of a short squeeze following a discriminatory auction tends to decrease with the auction size, increase with the market power of the largest long bidders, and decrease with small long players' scope for free-riding on a short squeeze. Asymptotically, as the auction size becomes arbitrarily large, the two types of auctions lead to equivalent outcomes.

**Keywords**: Multiple unit auction, uniform auction, discriminatory auction, treasury auction, repo auction, short squeeze, market manipulation, market power

**JEL**: D44, G12, G20, D62

This paper has been presented at the EuroConference on "Auctions and Market Design: Theory, Evidence and Applications" organised by Fondazione Eni Enrico Mattei and sponsored by the EU, Milan, September 26-28, 2002. The authors would like to thank the following people for helpful discussions and comments: Francesco Cafagna, Laurent Clerc, Steinar Ekern, Jan Ericsson, Julian Franks, Laurent Germain, David Goldreich, Francisco Gomes, Denis Gromb, Jűrgen von Hagen, Martin Hellwig, Anthony Neuberger, Mark Painting, and Raman Uppal. The authors also thank seminar participants at Bergen, Bocconi, Bologna, European Business School, London Business School, Mannheim, Padova, and the following conferences: Central Bank Operations: Theory and Evidence (organized by ZEI and the Bundesbank, September 2000), CEPR European Summer Symposium in Financial Markets (Gerzensee, July 2001), CEPR conference in Tenerife (May 2001). The usual disclaimer applies.

Address for correspondence: Kjell G. Nyborg London Business School Regent's Park London NW1 4SA United Kingdom Phone: +44 (0)20 7262 5050, Fax: +44 (0)20 7724 3317 E-mail: knyborg@london.edu

# 1 Introduction

The problem of how to organize the sale of many identical units is often solved in practice by holding an auction where bidders can submit multiple bids for multiple units. Such auctions are important, not least because the auctioned assets often play prominent roles in the wider economy. Examples include auctions of treasury securities, electricity, gold, and money. The scale of these auctions and the frequency with which they are held add to their importance. The feature of these auctions that we focus on in this paper arises from the fact that in practice bidders often have established forward positions in the underlying asset before the auction is actually held. Players that are short must cover their positions either by buying in the auction or the post-auction market. The risk of leaving it to the post-auction market lies in the chance that a few bidders buy so much of the underlying asset in the auction that they obtain market power in the secondary market. This power can be used to ask exorbitant prices when short players come to buy, i.e., to implement a short squeeze. This is known as the loser's nightmare (Simon, 1994). In this paper, we contribute to the theory of multiple unit auctions by showing how a potential short squeeze impacts on bidders' strategies and auction performance. We model the two most commonly used multiple unit auction formats, namely uniform price and discriminatory price auctions.

To get a perspective on the importance of the problem we are studying, in 2000 the US Treasury held 145 treasury auctions with a total nominal value of \$2.1 trillion. Moreover, primary dealers (who must bid in the auctions) often enter these auctions with substantial short positions as a result of pre-auction demand for the to-be-issued security by pension funds and other institutions [*Joint Report* (1992), Bikhchandani and Huang (1993), Simon (1994), Nyborg and Sundaresan (1996)]. Empirical evidence by Sundaresan (1994) suggests that short squeezing is a regular feature of this market. Direct evidence where a primary dealer bought most of the auction and subsequently squeezed the shorts is provided by the notorious Salomon squeeze:<sup>1</sup>

... the two-year notes became so scarce that the dealers who owned the notes charged exorbitant fees and financing costs when lending them to short-sellers. From small bond arbitrage operations in Chicago to the New York powerhouses, bond traders across America were badly burned. 'The arbs were hurt the worst; several of the smaller shops went out of business.'... The pain was so severe and the cries of foul play so loud that the two-year note squeeze became the talk of the bond market for weeks. (*Wall Street Journal*, August 19, 1991)

While this auction was discriminatory, since October 1998 all US Treasury auctions have been uniform.

Another important example of multiple unit auctions is repo auctions, which are used, for instance, by the European Central Bank (ECB) to channel euro denominated liquidity into the banking sector. ECB repo auctions are held every week and the typical size

<sup>&</sup>lt;sup>1</sup>See the Joint Report on the Government Securities Market (1992). Jegadeesh (1993) and Jordan and Jordan (1996) provide further discussion.

is around 100 billion euros. Since July 2000, they have been discriminatory.<sup>2</sup> In these auctions, financial institutions submit bids for how much they would like to borrow from the central bank at a given interest rate. The maturity of the loan is announced by the central bank in advance. In return for the funds, the financial institution hands over collateral to the central bank, which is returned when the loan is repaid. Some banks participating in repo auctions have a liquidity shortfall, perhaps as a result of their normal day to day activities. If these banks do not manage to cover their liquidity shortfall in the auction, they must borrow in the interbank market, often on an overnight basis. That a squeeze on liquidity can occur is suggested by the spikes observed in interbank rates around the end of the reserve maintenance period (see Hartmann, Manna, and Manzanares (2001) for European evidence and Hamilton (1996) for US evidence).

The possibility of a short squeeze in the post-auction market has importance for bidders in the auction as well as for the seller. From a short bidder's perspective, an important question is how to bid to avoid being squeezed. Similarly, a long bidder is interested in how to bid in order to implement a squeeze, or potentially free-ride off somebody else who will implement a squeeze. From a seller's perspective, revenue may be larger when the chance of a squeeze is larger, since this is likely to involve more aggressive bidding. Hence, a seller may view an auction procedure which promotes squeezing as desirable. However, a short squeeze also means that prices (or interest rates) are being distorted away from their competitive levels and experience higher volatility. This may be undesirable for many sellers such as sovereign treasuries and central banks. For example, the US Treasury has expressed a low award concentration as an auction objective and, to that end, do not allow individual dealers to buy more than 35% in the auction.<sup>3</sup> Whatever the seller's objective may be, it is important to establish the extent to which short squeezing and price distortion depend upon the auction format, size of the auction, pre-auction allocations and to what extent this is related to the seller's revenue.

To address these issues, we develop a model where a multiple unit auction of a homogeneous asset is followed by post-auction trading. Moreover, at the time of the auction, bidders may already have long or short positions in the to-be-auctioned asset. Since our aim is to focus on the interaction between strategic behavior in the post-auction market and the auction itself, pre-auction allocations are exogenously given. Short squeezing may happen in the post-auction market if some bidders are so large that they have market power. In both uniform and discriminatory auctions, bidders compete by simultaneously submitting collections of bids. Since the model is cast in the context of borrowing (or repos), individual bids consist of a quantity that the bidder wishes to borrow and an interest rate. But our analysis and results apply equally and in full to treasury auctions and,

 $<sup>^{2}</sup>$ Until June 2000, the ECB used to conduct repo auctions as fixed rate tenders. The interaction between short squeezing in the interbank market and bidder behavior in fixed rate tenders has been studied by Nyborg and Strebulaev (2001).

<sup>&</sup>lt;sup>3</sup>The 35% rule takes into account bidders' when-issued positions. As illustrated by the Salomon scandal, the US Treasury will punish dealers that attempt to get around this rule. The *Joint Report* (p.C-7) informs us that Salomon was prohibited from bidding in US Treasury auctions on behalf of its customers for an indeterminate time. Additionally, Salomon was fined nearly \$300 million (Sundaresan, 1997, p.72). In treasury auctions in other countries, e.g. Sweden, there are no such limits and indeed it happens from time to time that a single bidder buys the entire auction (Nyborg, Rydqvist and Sundaresan, 2001).

more generally, to security or commodity auctions where players buy the underlying asset outright and bids are price-quantity pairs. In both uniform and discriminatory auctions, the bids with the largest interest rates are hit first, until supply is exhausted. The difference between the two auction formats lies in the rate that is paid. In uniform auctions, all bidders pay the same "market clearing" rate; whereas in discriminatory auctions, bidders must pay the rate that they bid. This corresponds to how these types of auctions work in practice.

Our first finding deals with the valuation of the auctioned assets. In principle, our model is cast in a common value setting since in the absence of a squeeze, the competitive rate (or price) will prevail in the secondary market. However, we show that the possibility of a squeeze introduces a fundamental asymmetry between short and long bidders as well as between different types of long players, which means that they value the auctioned assets differently. In particular, short bidders tend to have downward sloping valuation schedules; whereas large long players tend to have upward sloping valuation schedules. Small long players tend to have flat valuation schedules. These differences in valuations across bidders as well as the varying marginal valuations for a single bidder has important consequences for equilibrium bidding strategies.

We find that, in equilibrium, short squeezes are more likely after discriminatory auctions than after uniform auctions. As a result, discriminatory auctions lead to more secondary market distortions, meaning deviations from the competitive rate (or price), in both the auction itself and in the secondary market. Another consequence is that volatility in the secondary market tends to be larger after discriminatory auctions. This is consistent with the empirical findings of Nyborg and Sundaresan (1996) in a study of the US experiment with uniform versus discriminatory price auctions in the 1990's. Finally, expected revenue is larger in discriminatory auctions than in uniform auctions.<sup>4</sup>

These results are driven only by the effect of a potential short squeeze. We do not consider the relative merits of uniform versus discriminatory auctions on other dimensions such as the winner's curse (see, e.g., Milgrom and Weber, 1982) or the extent to which they may lead to monopsonistic market power among bidders [Wilson (1979), Kyle (1989), Back and Zender (1993)]. Our paper draws on the literature on short squeezing, particularly on Dunn and Spatt (1984) and Cooper and Donaldson (1998). However, our main emphasis is on multiple unit auctions, and we are not familiar with any other model which examines the impact of a potential short squeeze on equilibrium in *auctions where bidders can submit multiple bids for multiple units*. Perhaps the most related work to ours is Chatterjea and Jarrow (1998), who model a pre-auction forward market, a single bid auction, and a post-auction market in which a short squeeze may occur.<sup>5</sup> However, our paper differs

<sup>5</sup>Other models of market cornering and short squeezing include Kyle (1984), Jarrow (1992 and 1994) and Kumar and Seppi (1992). These papers do not model auctions. Models of treasury auctions with

<sup>&</sup>lt;sup>4</sup>With respect to auction revenue, one take on the empirical evidence, e.g. as presented by Nyborg and Sundaresan (1996), on the performance of US Treasury auctions suggests that uniform auctions have larger revenue than discriminatory auctions. This is based on looking at the "markup" (the difference between the auction and when-issued rates before the auction). However, as pointed out by Nyborg and Sundaresan, one must interpret this difference with caution since, if a squeeze is more likely with a discriminatory auction, the "squeeze premium" is likely to be both in the auction rate and the when-issued rate. Hence revenue may be larger under discriminatory auctions even though the markup is larger.

substantially from theirs on several important dimensions. For example, (i) we model true multiple unit auctions, (ii) there can be any number of strategic players, and (iii) all players can participate in the auction regardless of their pre-auction position in the underlying asset. While Chatterjea and Jarrow study the case that a dealer attempts to squeeze a short player who cannot participate in the auction, we study how short and long players compete in the auction in the face of a potential short squeeze, which is the most relevant scenario in many contexts. For example, in US Treasury auctions, many primary dealers are often short in the when-issued market and are also the main participants in the auction (Joint Report, 1992).

An important issue which is raised in the short squeezing literature is the extent to which "small" long players are able to free-ride on a short squeeze by a "large" long player. For example, Kyle (1984) posits that small long players would be able to sell all their units well above the competitive price before the short squeezer would be able to sell any units. This is formalized and generalized by Cooper and Donaldson (1998) to the case that all players are strategic. The ability of smaller players to free-ride on a squeeze is also well recognized by traders that we have talked to. This is related to the well known point, when there are positive externalities, that smaller players can do better (on a per unit basis) than larger players, because the latter will often internalize the externality [Olsen and Zechkauser (1966), Bergstrom, Blume, and Varian (1986)].<sup>6</sup> A famous example from the finance literature is the ability of small shareholders to free-ride on the monitoring efforts of a large shareholder (Shleifer and Vishny, 1986). However, using a similar framework to Cooper and Donaldson, Dunn and Spatt (1984) provides a short squeezing model without free-riding. The contrast arises because of differences in the market microstructures. This raises an important question; namely, how bidding in the auction is affected by different scopes for free-riding. To study this, we employ a generalized short squeezing model which nests the models of Dunn and Spatt and Cooper and Donaldson as special cases. Surprisingly, we find that the main qualitative features of equilibrium in the auction are not sensitive to the scope for free-riding. However, the equilibrium probability of a short squeeze is decreasing in the scope for free-riding.

Since we explicitly characterize equilibria in both discriminatory and uniform auctions under any pre-auction allocation and any auction size, we are able to derive empirical predictions regarding various features of equilibrium bidding strategies. In particular, we find that in uniform auctions, the threat of a short squeeze tends to induce short players to submit collections of bids with a higher mean rate (or price) and more dispersion than long bidders. While there are pure strategy equilibria for uniform auctions, only mixed strategy equilibria exist for discriminatory auctions, where players randomize over bids. In these equilibria, short bidders tend to submit collection of bids with a higher expected mean rate (or price) than long bidders.

Another feature of equilibrium is that most long players are not active in the auction.

either a when-issued or a resale market include Bikhchandani and Huang (1989) and Viswanathan and Wang (2000). These papers do not consider short squeezes. For a recent survey on auction theory, see Klemperer (1999).

<sup>&</sup>lt;sup>6</sup>Olsen and Zeckhauser (1966), for example, attribute the disproportionately large share of GDP spent on defense among larger NATO allies to this point.

The roots of this result can be found in the different valuations of the shorts and longs. Since shorts value the first few units higher than the longs, they bid so aggressively for these units that most longs do not find it worthwhile to try to compete. This extreme behavior is quite surprising, particularly when the scope for free-riding is large, since a small long player would not lose his ability to free-ride in the secondary market if he were to buy a small amount in the auction.

Finally, under discriminatory auctions, the equilibrium probability of a short squeeze tends to decrease in the auction size, increase with the market power of the largest long players, and decrease with the scope for free-riding, *ceteris paribus*. As a result, price distortion, post-auction volatility, and revenue (per unit sold) tend to be smaller when the auction size is large (or market power is small). We also show that asymptotically discriminatory and uniform auctions are equivalent in that their outcomes converge to each other.

The rest of the paper is organized as follows. Section 2 describes the model. Section 3 analyzes the auction model when there are only two bidders. Section 4 generalizes and expands these results to any number of long bidders. This section also investigates the impact of the scope for free-riding on a squeeze. Section 5 draws out empirical predictions from the equilibria in Section 4. Section 6 discusses extensions to the model and Section 7 concludes. The appendix contains proofs not supplied in the text.

# 2 The Model

We construct a three date model where at date 1 there is a multiple unit auction, at date 2 there is trading in a secondary market, and at date 3 trades are settled and payoffs are collected. There are N players with initial positions in an underlying asset of  $\mathbf{Y}_0 = \{y_{n,0}\}_{n=1}^N \in \mathbb{Z}^n$ , where  $\mathbb{Z}$  denotes the set of integers. Initial allocations of individual players can be negative as well as positive. However, the total initial supply,  $Q_0 = \sum_{n=1}^N y_{n,0}$  is non-negative.<sup>7</sup> Initial allocations are common knowledge. We refer to players with negative (positive) positions as *short* (*long*).

We will think of the underlying asset as being money, and players with negative initial positions must refinance the loans that these positions represent by obtaining the required funds either by borrowing in the auction at date 1 or in the secondary market at date 2. All loans made at these two dates mature at date 3. The auction is held by an agency that does not participate actively in the game. This agency can be thought of as the central bank (or treasury), and the N primary players can be thought of as commercial (or investment) banks. One could equally think of the underlying asset as being a commodity or security

<sup>&</sup>lt;sup>7</sup>The assumption that  $Q_0 \ge 0$  is realistic, but not essential.  $Q_0 < 0$  would mean that some units of the physical underlying asset or long forward contracts on it would be held by players outside the model (see Section 6). The assumption that initial allocations are integer quantities corresponds to the fact that, in multiple unit auctions in practice, there is usually a quantity multiple. For example, in US Treasury auctions, individual bids must be for quantities in increments of \$1,000 nominal. In ECB repo auctions, the quantity multiple is 0.1 million Euros. In addition to being realistic, this discreteness assumption means that the type of underpricing equilibria in the uniform auction studied by Wilson (1979) and Back and Zender (1993) does not exist in our model (see Nyborg, 2001).

and the auction as being a *reverse repo auction* of the asset. In this case, players would be bidding to borrow the commodity until date  $3.^8$ 

For the purpose of describing a player's objective function, denote the award to the nth bank in the auction by  $y_{n,1}$  and the quantity weighted average interest rate that the bank must pay on this amount by  $a_{n,1}$ . Let  $y_{n,2} = y_{n,0} + y_{n,1}$  denote a bank's holdings at the beginning of date 2. Banks with  $y_{n,2} < 0$  must borrow at date 2 to cover their short positions. This opens up the possibility of short squeezing.<sup>9</sup> Denote the quantity weighted average interest at which the nth bank lends or borrows at date 2 by  $a_{n,2}$ . Thus, the total interest earnings or payments to bank n at date 3 are

$$\pi_n = a_{n,2}y_{n,0} + (a_{n,2} - a_{n,1})y_{n,1},\tag{1}$$

where all variables except  $y_{n,0}$  will be determined endogenously in equilibrium. The objective of a player is to maximize  $\pi_n$ .<sup>10</sup>

#### Date 1: Uniform and Discriminatory Auctions

The size of the auction is  $Q \in \mathbb{Z}_+$ . A bidder can make any number of bids such that the total quantity he demands is less than or equal to Q. An individual bid is an ordered pair  $(r,q) \in [R_l, \infty) \times \Omega$ , specifying an interest rate and a quantity, respectively, where  $\Omega = \{1, \ldots, Q\}$  and  $R_l$  represents the central bank's reservation rate.

Denote the set of bids submitted by player n by

$$\boldsymbol{b}_n = \{ (r_{n,i}, q_{n,i}) \}_{i=1}^{m(n)}, \tag{2}$$

where m(n) is the total number of bids submitted by the player. For the purpose of detailing the pricing and allocation rules in the two auction mechanisms, these bids can be ordered into a demand function  $x_n(r) = \sum_{i=1}^{m(n)} q_{n,i} \mathbb{1}_{[r_{n,i} \ge r]}$ , which is a left continuous, decreasing step function. The aggregate demand schedule is  $X(r) = \sum_{n=1}^{N} x_n(r)$ . The stop out rate,  $r_s$ , is the highest rate at which supply is exhausted (or  $R_l$  if no such rate exists). Specifically, since X(r) is a left continuous step function,

$$r_s = \begin{cases} \max\{r|X(r) \ge Q\} & \text{if } \{r|X(r) \ge Q\} \neq \emptyset \\ R_l & \text{otherwise.} \end{cases}$$

The auctioned supply is allocated to the highest bids. This means that bids above the stop-out rate are awarded in full, while bids at the stop-out rate are rationed (pro rata). To formalize this, let  $dx_n(r_s)$  be the marginal demand of player n at the rate  $r_s$  and let

<sup>&</sup>lt;sup>8</sup>If the security or commodity were purchased outright instead of through a repo, algebraically and economically, the model and the results would be unaffected (see footnotes 9 and 10). In the context of when-issued markets and treasury auctions, date 2 would represent the issue date.

<sup>&</sup>lt;sup>9</sup> In the case that the underlying asset is a security or commodity, short squeezing requires that the asset is unique in the sense that it is impossible to cover short positions by close substitutes. For example, in US Treasury when-issued markets, only securities with particular CUSIPs can serve this function.

<sup>&</sup>lt;sup>10</sup> In the case that the underlying asset is a security or commodity and is purchased outright instead of borrowed through a reverse repo, the objective function (1) can also be interpreted in terms of prices. In this case,  $a_{n,t}$  would be a quantity weighted average price.

 $x_n(r_s^+)$  denote his demand at prices above  $r_s$ .<sup>11</sup> Then the *n*th bank receives an auction award of

$$y_{n,1} = x_n(r_s^+) + \frac{dx_n(r_s)}{\sum_{i=1}^N dx_i(r_s)} \left[ Q - \sum_{i=1}^N x_i(r_s^+) \right].$$
 (3)

The difference between uniform and discriminatory auctions lies in the rate that winning bidders must pay. In a uniform auction, all winning bidders pay the stop-out rate; while in a discriminatory auction, winning bidders pay what they bid. Hence, for uniform auctions, the interest costs on the winning bids of the nth bank will be

interest in uniform auction  $= a_{n,1}y_{n,1} = r_s y_{n,1}$ .

For discriminatory auctions, the nth bank's interest costs is the sum of the interest on its winning bids, that is,

interest in discriminatory auction 
$$= a_{n,1}y_{n,1} = \sum_{i=1}^{m(n)} q_{n,i}r_{n,i}\mathbf{1}_{[r_{n,i}>r_s]} + [y_{n,1} - x_n(r_s^+)]r_s.$$

Hence for discriminatory auctions  $a_{n,1}$  can be larger than the stop-out rate.

#### Date 2: The Secondary Market and Short Squeezing

Although the primary focus of our analysis is on the auction itself, what makes this new and interesting is the possibility of short squeezing in the secondary market. To model this, we employ a reduced form representation which nests the models of Dunn and Spatt (1984) and Cooper and Donaldson (1998) as special cases. These models share a common framework which, for our purposes, has two important features; namely, all players are strategic and multiple players can have sufficient market power to implement a short squeeze. In our context, this framework can be described as follows:

Date 2 starts with longs making offers to shorts to lend the underlying asset for one period (until date 3). Shorts also have the outside option of borrowing from the central bank, say, at  $R_h$ . This rate therefore caps what a long player can earn from a squeeze. After shorts have refinanced, longs can lend any remaining units of the underlying asset, perhaps to retail clients, at the competitive rate of  $R_0 \in [R_l, R_h)$ .<sup>12</sup> There is a short squeeze in equilibrium if and only if some player has market (monopoly) power, which using our notation is defined as follows:

**Definition 1** Let  $\mathcal{N}$  denotes the set of all N players. For  $t \in \{0, 2\}$ , the **market power** of player n is

$$z_{n,t} \equiv \max\left[0, -\sum_{i \in \mathcal{N}/n} y_{i,t}\right].$$
(4)

<sup>11</sup> Formally, (i)  $x_n(r_s^+) = \sum_{i=1}^{m(n)} q_{n,i} \mathbb{1}_{[r_{n,i} > r_s]}$ , and (ii)  $dx_n(r_s) = \sum_{i=1}^{m(n)} q_{n,i} \mathbb{1}_{[r_{n,i} = r_s]}$ , so  $dx_n(r_s) = 0$  if bidder *n* places no bids at  $r_s$ .

<sup>&</sup>lt;sup>12</sup>In the contexts of money markets,  $R_h$  would represent the central bank's lending standing facility. In most countries or currency areas, the competitive rate is typically straddled by the lending and deposit standing facilities of the central bank. More generally,  $R_h$  could represent a rate between the competitive rate and the lending standing facility, perhaps determined by bargaining. In other contexts,  $R_h$  would represent the price of the "fancy good" (see, e.g., Salant (1984)). In treasury auctions  $R_h$  would be determined by the penalty from failing to deliver on a trade from the when-issued market. This penalty may differ between countries.

This says that at date 2, the market power of the *n*th bank is the units of the underlying asset held by that bank which the shorts need to cover their positions and cannot obtain elsewhere (without using the outside option). Shorts and "small" longs have no market power. If no player has market power, all units trade at the competitive rate of  $R_0$ . However, if a player has market power over z units, he can lend these units at  $R_h$ .

The question that remains is: In the case of a short squeeze, what is the transaction rate for those units that shorts need to cover and over which no player has monopoly power? As it turns out, there is no consensus answer in the literature. In Dunn and Spatt (1984), the equilibrium transaction rate on these units is  $R_0$ . In contrast, in Cooper and Donaldson (1998), when there is a unique largest long player, all other longs are able to lend all their units at  $R_h$ , as suggested by Kyle (1984). In other words, in equilibrium small players free-ride on the squeeze of a large long player, and shorts get squeezed on all units they need to cover. The different conclusions reached by these papers is due to differences in trading mechanisms. The general point seems to be that the extent to which "small" longs are able to free-ride on a squeeze depends on how the market is organized.

To study the impact of different levels of free-riding, we introduce  $\delta \in [0, 1]$  as a measure of the scope for free-riding, where  $\delta = 0$  denotes no free-riding opportunities (Dunn and Spatt) and  $\delta = 1$  denotes full free-riding opportunities (Cooper and Donaldson). Now, there are three types of longs at date 2. First, there are the small longs, who have zero market power. Second, there are the intermediate longs, who have positive market power but not the largest. Third, there are the X largest longs who have the largest positive market power. We denote these by  $L_0$ ,  $L_1$ , and  $L_2$ , respectively. If no player has market power, X = 0 and  $L_1$  and  $L_2$  are empty. Given the history of the game up to date 2, payoffs to long players are as follows:<sup>13</sup>

$$\pi_n = \begin{cases} [R_0 + \delta \mathbf{1}_{[X \ge 1]} (R_h - R_0)] y_{n,2} - a_{n,1} y_{n,1} & \text{if } n \in L_0 \\ R_h z_{n,2} + [R_0 + \delta (R_h - R_0)] [y_{n,2} - z_{n,2}] - a_{n,1} y_{n,1} & \text{if } n \in L_1 \\ R_h z_{n,2} + R_0 (y_{n,2} - z_{n,2}) - a_{n,1} y_{n,1} & \text{if } n \in L_2. \end{cases}$$
(5)

Notice that when  $\delta$  is relatively large, small and intermediate longs do better on a per unit basis than the largest longs and, in some cases, may even do better in absolute terms.

The aggregate payoff to the shorts is determined by (5) and the fact that the total gross payoff to all players (payoff net of interest to be paid from units obtained in the auction) is

$$(Y_L - Y_S)R_0, (6)$$

where  $Y_L = \sum_{n \in \mathcal{N}_2^+} y_{n,2}$ ,  $Y_S = \sum_{n \in \mathcal{N}_2^-} |y_{n,2}|$ , and where  $\mathcal{N}_t^+ = \{i | y_{i,t} \ge 0\}$  denotes the set of long players and  $\mathcal{N}_t^- = \{i | y_{i,t} < 0\}$  denotes the set of short players at date t. Define

<sup>&</sup>lt;sup>13</sup>In their analysis, Cooper and Donaldson consider only the case that there is at most one player with market power. A proof that the general case (when many longs have market power) yields payoffs as stated in (5) with  $\delta = 1$  is available from the current authors upon request. Note that we are focusing on their "endgame/delivery process," since, for our purposes, the dynamic aspects of their model is of secondary importance. Dunn and Spatt's model is only developed for one short. However, the general point is that there may be a trading mechanism which delivers the competitive price on those units over which players do not have monopoly power. In a more standard, non-squeeze setting, Allen and Hellwig (1986) have shown in a model where capacity constrained sellers choose prices as strategies, that the equilibrium price converges in distribution to the competitive price as the number of sellers increases.

 $Y_0 = \sum_{n \in L_0} y_{n,2}, Y_1 = \sum_{n \in L_1} (y_{n,2} - z_{n,2})$ , and  $Z_L = \sum_{n \in \mathcal{N}_2^+} z_{n,2}$ . By subtracting from (6) the gross payoffs to the various long players using (5) and adding the shorts' interest costs from units obtained in the auction, we find that the aggregate payoff to the shorts is:<sup>14</sup>

$$\sum_{n \in \mathcal{N}_2^-} \pi_n = -R_h Z_L - R_0 (Y_S - Z_L) - \delta \mathbb{1}_{[X \ge 1]} (R_h - R_0) (Y_0 + Y_1) - \sum_{n \in \mathcal{N}_2^-} a_{n,1} y_{n,1}.$$
(7)

The first term in (7) represents the squeeze itself on  $Z_L$  units. The second term is what the shorts would pay to cover the remaining units that they are short, if there were no free-riding. The third term represents the free-riding. This says that the number of units that no player has market power over and on which the shorts pay the squeeze premium,  $R_h - R_0$ , is equal to  $Y_0 + Y_1$ . In other words, it is equal to the sum of the positions of all small longs and the sum of the positions of all intermediate longs in excess of their market power.

Finally, notice that (4) also defines market power at date 0. This is essentially what date 2 market power would be if no units were auctioned at date 1. To retain his entire date 0 market power, a long player will need to buy all units in the auction. To guarantee that no player will have market power at date 2, it is sufficient for the shorts to buy a total of

$$Z \equiv \max\{z_{n,0} | n \in \mathcal{N}\}$$

units in the auction. This parameter will play a central role in the auction analysis.

## 3 The Auction: Equilibrium with Two Players

By studying the model with only two players, one short and one long, we are able to draw out some of the important differences between uniform and discriminatory auctions while maintaining a fairly simple modeling structure. The additional complexities raised by having more than two long players are studied in subsequent sections.

As a benchmark, observe that if both players have long positions initially, the auction rate is  $R_0$ . In equilibrium under either auction format, both banks submit bids for Q units at a rate of  $R_0$ . Therefore, in this section, we study the more interesting case that one player is short and the other is long. Let  $y_{1,0} < 0$  and  $y_{2,0} = Q_0 - y_{1,0} > 0$ . Hereafter we will refer to Player 1 as "the short (player)" and to Player 2 as "the long (player)". Since there are only two players, the market power of the long player at date 0 is  $Z = -y_{1,0}$ .

For a given outcome in the auction,  $\{y_{1,1}, y_{2,1}, a_{1,1}, a_{2,1}\}$ , equations (5) and (7) tell us that trading in the secondary market will yield the following date 3 payoffs:

$$\pi_n = \begin{cases} y_{1,2}R_0 - z_{2,2}(R_h - R_0) - a_{1,1}y_{1,1} & \text{if } n = 1. \\ y_{2,2}R_0 + z_{2,2}(R_h - R_0) - a_{2,1}y_{2,1} & \text{if } n = 2. \end{cases}$$
(8)

<sup>&</sup>lt;sup>14</sup> In deriving (7), we have used the following implications of the definition of market power: (i) for any player with market power at date 2,  $y_{n,2} - z_{n,2} = Y_L - Y_S$ ; and (ii)  $Z_L = Y_L - Y_0 + |L_2|(Y_S - Y_L) + |L_1|(Y_S - Y_L)$ , where  $|L_1|$  is the number of intermediate longs. Note also that (ii) implies that for  $X \ge 1$ ,  $Y_1 + Y_0 = Y_S - Z_L - (X - 1)(Y_L - Y_S)$ . This shows that the aggregate payoff to the shorts at date 2 is increasing in X when  $\delta > 0$ , ceteris paribus. This is a feature of Cooper and Donaldson's model. Details are available from the current authors upon request.

where we have used  $z_{2,2} = \max[0, -y_{1,2}]$ . The expression for  $\pi_2$  can be rewritten as  $R_0y_{2,0} + (R_h - R_0)z_{2,2} - (a_{2,1} - R_0)y_{2,1}$ , which shows that Player 2 faces a potential pricequantity tradeoff in the auction from squeezing in the secondary market: he gains  $R_h - R_0$  for each unit of market power he has at date 2, but loses  $a_{2,1} - R_0$  for each unit purchased in the auction. A similar tradeoff is seen in the expression for  $\pi_1$ .

We refer to an auction as *small* if  $Q \leq Z$ . In this case, there will always be a short squeeze at date 2 under any auction format, since the auction does not offer the chance for the short player to cover his position. In this section, we focus on the more interesting case that the auction is *large* in the sense that Q > Z, which we believe is the most relevant scenario in practice. In this case, the short can cover in the auction, if he bids sufficiently aggressively. As a result, it is not a foregone conclusion that there will be a squeeze.

#### 3.1 Private Valuations in Large Auctions

The setting in our paper is seemingly common value since, in the absence of a short squeeze, the prevailing rate is  $R_0$  for all units. Moreover, in the event of a squeeze the extra interest paid by a short bank goes directly to a long bank. However, we show here that there is a simple but fundamental asymmetry between short and long players which leads them to value the auctioned units differently.

To illustrate the difference in valuations, we ask the following question: how much is Player *n* willing to pay to obtain an additional unit in the auction, keeping the total number of units in the auction fixed at Q > Z. We refer to the schedule of these net marginal valuations as the player's valuation schedule,  $V_n$ . Thus  $V_n(q) = a_{n,2}^*(q)(y_{n,0} + q) - a_{n,2}^*(q-1)(y_{n,0} + q - 1)$ , where  $a_{n,2}^*(q)$  is the date 2 equilibrium quantity weighted average rate at which the *n*th player lends or borrows his  $y_{n,0} + q$  units.

**Proposition 1** The valuation schedules of the auctioned units of the two players are different. In particular, the short has a decreasing valuation schedule, and the long has an increasing valuation schedule.

**Proof:** To avoid being squeezed, the short needs to win only Z units in the auction. Since any additional units that the short wins can be lent at  $R_0$ , the short values the first Z units he wins in the auction at  $R_h$  and the last Q - Z units at  $R_0$ . The long needs to win Q - Z + 1 units in the auction in order to implement a squeeze. Therefore, the long values the first Q - Z units he wins in the auction at  $R_0$  and the last Z units at  $R_h$ .  $\Box$ 

This illustrates the fundamental result that the possibility of short squeezing in the secondary market can give rise to differential valuations in the primary market in a seemingly common value setting. This breakdown of a common valuation, both across and within bidders' valuation schedules, impacts profoundly on our analysis and is the reason why, as we shall see, uniform and discriminatory auctions lead to different outcomes.

The idea that a potential short squeeze can give rise to private valuations in multi-unit auctions was first suggested by Sundaresan (1994). Our setting is different from the private values model studied by Vickrey (1961) and others in that: (i) in our model, shorts have decreasing valuations schedules while longs tend to have increasing valuation schedules, and (ii) these valuation schedules are consequences of equilibrium in the secondary market and are not independent of each other.

Coming back to the valuation of the long, we see that the most he would be willing to pay for Q units in the auction is (per unit)

$$\bar{R}_{QZ} \equiv \frac{(Q-Z)R_0 + ZR_h}{Q} < R_h.$$
(9)

 $R_{QZ}$  is the quantity weighted average interest rate the long would earn on the Q auctioned units, if he were to win them all. This average, or break-even, value turns out to be an important parameter in the auction analysis.

#### 3.2 Large Uniform Auctions

Recall that a pure strategy for Player n in the auction specifies a set of bids, as represented by (2) with  $\sum_{i=1}^{m(n)} q_{n,i} \leq Q$ , as a function of the player's initial allocation. The equilibrium concept is Nash equilibrium in either pure or mixed strategies.

**Theorem 1** In uniform auctions, it is equilibrium for Player 2 to submit  $\mathbf{b}_2^u = \{(R_0, Q)\}$ and for Player 1 to submit  $\mathbf{b}_1^u = \{(R_h, Z), (R_0, Q - Z)\}$ . The outcome of this equilibrium, which is also the unique equilibrium outcome, is as follows: First, the revenue to the seller is  $QR_0$  and the stop-out rate is  $R_0$ . Second, payoffs to the two players are given by  $\pi_n = y_{n,0}R_0$ , n = 1, 2. Third, there is no short squeeze in the secondary market.

Intuitively, the short is exploiting the differential valuations by bidding for the units he needs to cover at a very high price. Faced with this, the long can do no better than being passive (placing no bids above  $R_0$ ). Thus, there is no cost to the short from bidding so aggressively since he only pays the stop-out rate for all units he wins. Hence, the combination of differential valuations and uniform price eliminates short squeezing. As a consequence, all units in the auction are sold at the competitive rate, and there is no interest rate distortions or excess volatility in the secondary market.

### 3.3 Large Discriminatory Auctions

The analysis for discriminatory auctions is more complicated because there is no equilibrium in pure strategies. This is more than a technical result. It also implies that in any equilibrium, there is a positive probability of a post-auction short squeeze.

**Lemma 1** There is no equilibrium in pure strategies in discriminatory auctions.

This is a consequence of the different marginal valuations outlined above and the fact that bidders in discriminatory auctions pay what they bid. To see this, suppose for example that the short submits bids for Z units at  $R_h$  and Q - Z units at  $R_0$ , as he does in the uniform auction. The best response for the long would be to submit a bid for Q units, say, at  $R_0$ . But then the short could do better by demanding Z units at a rate marginally above  $R_0$ , to avoid being squeezed. But then the long could improve by bidding more aggressively, etc. **Theorem 2** In discriminatory auctions, the following is an equilibrium:

(i) The short submits  $\mathbf{b}_1^* = \{(\tilde{S}, Z), (R_0, Q - Z)\}$ , where  $\tilde{S}$  is a random variable with realization S, support  $S \in [R_0, \bar{R}_{QZ}]$ , and cumulative distribution function

$$F(S) = \frac{Q - Z}{Z} \frac{S - R_0}{R_h - S}.$$
 (10)

(ii) The long submits  $\mathbf{b}_2^* = \{(\tilde{L}, Q)\}$ , where  $\tilde{L}$  is a random variable with realization L, support  $L \in [R_0, \bar{R}_{QZ}]$ , and cumulative distribution function

$$G(L) = \frac{Q - Z}{Q} \frac{R_h - R_0}{R_h - L} = \frac{R_h - \bar{R}_{QZ}}{R_h - L}.$$
(11)

The first part of the proof is interesting because it shows how F(S) and G(L) are constructed, and we therefore include it here (the second part is in the appendix).

**Proof:** We need to show that  $b_1^*$  and  $b_2^*$  are best replies to each other. Observe first that under these strategies, each player demands Q units at a price of  $R_0$  or higher with probability one. Therefore, since the short values the first Z units at  $R_h$  and the last Q - Z units at  $R_0$ , he cannot do better than letting one of his bids be  $(R_0, Q - Z)$ . Furthermore, neither bidder can improve on his payoff by deviating from the proposed equilibrium by submitting a bid where the rate has a positive probability of being below  $R_0$ .

Observe next that the proposed F(S) has no mass points. Therefore, under the proposed strategies, either the long will win Q units and squeeze on Z units, or he will win fewer than Qunits and squeeze on no units. In the former case, the payoff to the long is  $\pi_L = (Q - Z)R_0 + ZR_h - QL \ge 0$  if and only if  $L \le \bar{R}_{QZ}$ , by definition of  $\bar{R}_{QZ}$ . Hence, the long cannot do better than having  $G(\bar{R}_{QZ}) = 1$ . Furthermore, since the proposed G(L) does not have a mass point at  $\bar{R}_{QZ}$ , the short cannot do better than having  $F(\bar{R}_{QZ}) = 1$ . Hence, neither bidder can improve his payoff by expanding the proposed supports of F(S) or G(L).

Since F(S) and G(L) are continuous CDF's on  $[R_0, R_{QZ}]$ , the short's expected payoff is

$$E[\pi_S] = \int_{[R_0, \bar{R}_{QZ}]} (1 - G(S))(-ZR_h) + G(S)(-ZS) \, dF(S), \tag{12}$$

and the long's expected payoff is

$$E[\pi_L] = (y_{2,0} + Q - Z)R_0 + \int_{[R_0, \bar{R}_{QZ}]} F(L)(ZR_h - QL) + (1 - F(L))(Z - Q)L\,dG(L).$$
(13)

For the proposed strategies to be equilibrium, it must be the case that the integrands of these two expressions are independent of S and L, respectively. In particular, this means that the long's strategy must satisfy

$$(1 - G(S))(-ZR_h) + G(S)(-ZS) = C_1,$$
(14)

where  $C_1$  is a constant. Since  $G(\bar{R}_{QZ}) = 1$ , it follows that  $C_1 = -Z\bar{R}_{QZ}$ . From this, a bit of algebra shows that G(L) as given by (11) is the unique solution to (14). Similarly, the short's strategy must satisfy

$$F(L)(ZR_h - QL) + (1 - F(L))(Z - Q)L = C_2,$$
(15)

where  $C_2$  is a constant. Since  $F(\bar{R}_{QZ}) = 1$ , it follows that  $C_2 = (Z - Q)R_0$ , from where it is found that F(L) as given by (10) is the unique solution to (15). This establishes that the proposed strategies constitute equilibrium, provided that the bidders cannot improve their payoffs by splitting their bids further (see appendix).

In the discriminatory auction equilibrium described in Theorem 2, the short splits his bid while the long submits a single bid. This parallels the result for uniform auctions. As before, the intuition for why the short submits two bids derives from the fact that he values the first few units he needs to cover higher than the remaining auctioned units. But in the discriminatory auction, the randomness of the short's strategy affords the long with the opportunity to implement a short squeeze. The long takes advantage of this by submitting a bid for the entire auction at a single rate which is above  $R_0$  with positive probability. This reflects the facts that the long needs to buy all units to maximize the size of the squeeze and his valuation schedule is increasing.

Figure 1 illustrates the distribution used by the short, F(S), and the long, G(L), for Z and Q units, respectively. It is seen that F(S) first order stochastically dominates G(L). Intuitively, this happens because the short values the first Z units higher on average than the long values all Q units. Furthermore, in the event that there is no squeeze, the long ends up buying Q - Z units, which he values at only  $R_0$ . As a response to this, the long chooses G(L) to have a mass point at  $R_0$ , as seen in the figure. In other words, the long is aggressive (i.e., bids above  $R_0$ ) only part of the time.

The equilibrium in Theorem 2 is a member in the following class:

**Definition 2** An equilibrium is a single-bid equilibrium if at least one player submits only one bid (i.e., for some n,  $\mathbf{b}_n = \{(\tilde{r}_n, \tilde{q}_n)\}$ , where  $\tilde{r}_n$  and  $\tilde{q}_n$  are random variables).

This definition allows for both pure and mixed strategies. Single-bid equilibria are attractive because they are relatively simple. The equilibrium identified in the uniform auction is also a single-bid equilibrium. Moreover, it is outcome unique. Here, we establish a similar, but somewhat weaker result for the discriminatory auction.

**Theorem 3 (Uniqueness)** In discriminatory auctions, the equilibrium described in Theorem 2 is the unique single-bid equilibrium.

In the equilibrium in Theorem 2, the mixed strategies of the short and the long have support on  $[R_0, \bar{R}_{QZ}]$ , which is intuitive since  $\bar{R}_{QZ}$  is the average value to the long if he wins all Q units. Theorem 3 establishes that there is no single-bid equilibrium with different support.

#### **3.4** Market Entry in Large Auctions

In this subsection, we complete the analysis of large auctions with two players by asking whether our results are robust to market entry. The equilibria we have identified above are not affected by bids from third parties as long as these bids are placed at  $R_0$ . So we ask the question as to whether other players, who have zero initial allocations, are tempted to bid aggressively in the auction if that were possible with zero transaction costs. By "bidding aggressively" we mean placing bids in the auction above  $R_0$  with positive probability. **Proposition 2** Suppose the auction is uniform and the long and the short use the equilibrium strategies  $\{\mathbf{b}_1^u, \mathbf{b}_2^u\}$ . It is not optimal for a player with no endowment to bid aggressively in the auction (regardless of the scope for free-riding).

The intuition derives from the passive bidding of the long and the aggressive bidding of the short for Z units, which means that a short squeeze cannot happen regardless of how many units new entrants may obtain in the auction. For example, if new entrants buy the entire auction, none of them nor Player 2 will have market power in the secondary market.

**Proposition 3** Suppose the auction is discriminatory and the long and the short use the unique single-bid equilibrium strategies,  $\{b_1^*, b_2^*\}$ . It is not optimal for a player with no endowment to bid aggressively in the auction (regardless of the scope for free-riding).

It may seem more attractive for a potential new entrant to actually participate in a discriminatory auction, as compared with a uniform auction, since he may be able to free-ride on a squeeze. However, the proposition shows that even when the scope for free-riding is at the maximum, the potential gains from this are too small compared with the cost of bidding aggressively, winning a few units, and then not having a squeeze materialize because the short managed to cover. The intuition relates to the aggressiveness of both the short and the long, which is illustrated by the fact that the long's equilibrium expected profit is  $y_{2,0}R_0$ ; i.e., his efforts to implement a short squeeze does not earn him any abnormal returns (see the proof of Theorem 2). Furthermore, as discussed above, the short is even more aggressive than the long.

# 4 Auction Equilibrium with Several Long Players

This section expands the analysis by considering the case that there are several long players. We continue to assume that there is only one short player. Thus we address the question as to what are the effects of competition among potential short squeezers and what is the impact of the scope for free-riding. For example, does the short player or the auctioneer benefit from competition among many longs? The answer is not obvious because an individual long may bid passively in the auction, hoping to free-ride on somebody else's short squeezing efforts.

We have N players, such that the first is short and all others are long. Without loss of generality, let  $y_{2,0} \ge y_{3,0} \ge \ldots \ge y_{N,0}$ . The number of long players with positive market power at date 0 is denoted by K. So only players  $2, \ldots, K+1$  have market power initially. We will be interested in longs with the largest market power: let there be M of them,  $M \le K$ , with numbers  $n = 2, \ldots, M+1$ .

### 4.1 Valuation Schedules in Large Auctions with Many Longs

The basic message from the case of N = 2, that players with different initial endowments have different valuation schedules, remains true in the more general case we are considering here. However, the exact value of an additional unit to a player now depends upon how the remaining units are distributed among the other players. For example, the value to the short from capturing a qth unit in the auction is: (i)  $R_h$  if he will be squeezed regardless of whether he wins q - 1 or q units; (ii)  $R_0$  if he will not be squeezed; or (iii)  $R \ge R_h$  if by winning this extra unit the short goes from being squeezed to not being squeezed. To see this, suppose there are three players, initial allocations are  $\{-8, 12, 5\}, Q = 5$ , and  $\delta = 1$ . If the short captures 2 units and the largest long captures 3 units, the short will be squeezed to not being squeezed and can borrow 5 units at date 2 at  $R_0$ . Therefore, the value to the short of the third unit in this case is  $R = 6R_h - 5R_0 > R_h$ . Which of the three scenarios apply depends upon which players would get the remaining units. Thus it is now impossible to unequivocally specify the short's valuation schedule. However, because the same fundamental forces are at work here as when N = 2, the short generally values the first few units higher than the last unit, which he always values at  $R_0$ , since Q > Z.

For long players, we need to distinguish between "small" longs (players with no market power at date 0) and "large" longs (players with market power at date 0). This distinction is important because only large longs can implement a squeeze at date 2. (If a small long buys the entire auction, there will be no long players left with market power at date 2). The value to a large long from winning a qth unit is: (i)  $R_h$  if he will be implementing a short squeeze, (ii)  $R_0 + \delta(R_h - R_0)$  if he will free-ride on somebody else's squeezing efforts; (iii)  $R_0$ if none of the longs are sufficiently large to implement a squeeze (perhaps because all the other Q - q units are going to the short); or (iv)  $R \leq R_0$  if by winning this extra unit the player becomes just large enough that he stops somebody else implementing a squeeze. Again, we cannot unequivocally specify a large long's valuation schedule. However, in general, a large long values the first few units lower than the last unit, which he always values at  $R_h$ , since if he buys all units he will implement a short squeeze for sure.

Finally, the value to a small long from capturing a *q*th unit can also be  $R_0 + \delta(R_h - R_0)$  (if he free-rides on a squeeze),  $R_0$  (if there is no squeeze), or  $R \leq R_0$  (if he stops a squeeze). Unlike a large long, the small long generally values the first few units higher than the last unit, which he always values at  $R_0$ . This opposite pattern is a result of the small long's inability to implement a short squeeze.

The above discussion illustrates the conditional nature of valuation schedules. It also illustrates that the short has a strong incentive to bid aggressively for a few units only, in order to try to avoid being squeezed. Long players may also benefit from bidding aggressively for a few units only, provided that they can free-ride on somebody else's squeezing efforts. However, their incentive is not as strong as the short's, because of the risk that a short squeeze will not happen.

#### 4.2 Large Uniform Auctions with Many Longs

In uniform auctions, the fact that bidders pay the stop-out rate – as opposed to what they actually bid – is exploited by the short player, in the same way as for N = 2.

**Theorem 4** It is equilibrium for Player 1 to submit  $\mathbf{b}_1^u = \{(R_h, Z), (R_0, Q - Z)\}$  in uniform auctions and for all other players to submit  $\mathbf{b}_n^u = \{(R_0, Q)\}$  (regardless of the scope for free-riding). The outcome of this equilibrium, which is also the unique equilibrium

outcome, is as follows: First, the revenue to the seller is  $QR_0$  and the stop-out rate is  $R_0$ . Second, for every n, the payoff is given by  $\pi_n = y_{n,0}R_0$ . Third, there is no short squeeze in the secondary market.

As before, the short places a larger value than the longs on the first Z units. Therefore, he bids aggressively for these units and, in a sense, thus pre-empts a short squeeze. This aggressive bidding is not costly to the short since the absence of a squeeze means that longs are not willing to bid above  $R_0$ , implying that the equilibrium stop-out rate is  $R_0$ .

#### 4.3 Large Discriminatory Auctions with Many Longs

In large discriminatory auctions with many longs there is no pure strategy equilibrium, for the same reason as for N = 2. When the scope for free-riding is large, a potentially important force in discriminatory auctions is the disincentive for long players to bid aggressively, since smaller longs may do better on a per unit basis than larger longs.

**Theorem 5 (Existence, characterization, and uniqueness)** Suppose the auction is discriminatory. For each of the M long players with the largest market power at date 0, there is an equilibrium in which only that player and the short are aggressive, with all other players being passive. In particular, for each n = 2, ..., M+1 the following is equilibrium: (i) The short submits  $\mathbf{b}_1^* = \{(\tilde{S}, Z), (R_0, q_{1,2}^*)\}$ , where  $q_{1,2}^* \in \{0, ..., Q - Z\}$  and  $\tilde{S}$  is a random variable with realization  $S \in [R_0, \bar{R}_{QZ}]$  and cumulative distribution function

$$F(S) = \frac{Q - Z}{Z} \frac{S - R_0}{R_h - S}.$$
(16)

(ii) Player n submits  $\mathbf{b}_n^* = \{(\tilde{L}, Q)\}$ , where  $\tilde{L}$  is a random variable with realization  $L \in [R_0, \bar{R}_{QZ}]$  and cumulative distribution function

$$G(L;\delta) = \frac{Z(R_h - \bar{R}_{QZ}) + \delta(R_h - R_0)(|y_{1,0}| - Z)}{Z(R_h - L) + \delta(R_h - R_0)(|y_{1,0}| - Z)}.$$
(17)

(iii) Every other player i submits  $\mathbf{b}_i^* = \{(R_0, Q)\}$ . Finally, there is no other equilibrium in which only one long is aggressive and he or the short submits only one bid.<sup>15</sup>

This theorem describes M different equilibria, one for each player with the largest market power at date 0. Since these players are indistinguishable from each other, the M equilibria are observationally equivalent. In these equilibria, only the short and one of the largest longs are active. The two active banks play mixed strategies and a short squeeze occurs with positive probability. While the active long submits only one bid, the short splits his bids into a high bid for the units he needs to cover and a bid for some additional units at the competitive rate,  $R_0$ . The exact amount the short bids for at  $R_0$  has no effect on

<sup>&</sup>lt;sup>15</sup>To be precise, the equilibria described in the theorem are unique up to trivial variations such as where some passive bidders do not submit bids at all, demand less than Q at  $R_0$ , or submit bids below  $R_0$ , etc.

bidders' payoffs. In a sense, therefore, the theorem shows that there is a unique single-bid equilibrium also for N > 2.

The result that no other banks are active in equilibrium expands our "no-entry" result for N = 2. However, the current result is more general as it shows that, in equilibrium, most long players strategically choose not to participate actively in the auction. This is surprising. When the scope for free-riding is large, it is partly driven by the desire to free-ride on a squeeze. But this alone cannot explain why these bidders do not attempt to augment their positions in the auction by bidding aggressively for a small number of units. Furthermore, when the scope for free-riding is small, one might think that longs would compete in the auction in order to be the player with the largest market power at date 2. In either case, most longs decide not to bid aggressively for any number of units because the risk that the short will manage to cover in the auction is too large.<sup>16</sup>

Theorem 5 establishes that there are no "single-bid" equilibria where one of the smaller longs with market power is aggressive and all other longs are passive.<sup>17</sup> The intuition has its roots in how the short tailors his strategy according to the market power of the active long. The proof of the theorem shows that in a hypothetical equilibrium where the active long has  $z_{n,0} < Z$ , the short is not as aggressive as when one of the largest longs is active. The implication is that one of the largest longs can step in and earn abnormal profits. Finally, the theorem also demonstrates that the equilibrium impact of the scope for free-riding is quantitative but not qualitative. In particular, the only effect is on the distribution used by the aggressive long.

#### 4.4 Small Auctions with Many Longs

Here, we complete the analysis by considering small auctions.

**Theorem 6** Suppose Q < Z. Irrespective of the auction type and the scope for free-riding, in equilibrium: (i) There is a short squeeze for sure; and (ii) auction revenue is  $QR_h$ .

The intuition for why all units are lent at  $R_h$  irrespective of the auction format relates to the players' valuation schedules. When the auction is small, the short is unable to cover, implying that there will be a squeeze for sure and consequently that at least two players value each auctioned unit at  $R_h$ . Competition drives the auction rate up to  $R_h$ .<sup>18</sup>

In contrast with large discriminatory auctions, after a small auction has been held, several players may turn out to have the largest market power. This can happen, for

<sup>&</sup>lt;sup>16</sup>Jehiel and Moldovanu (1996) show, in a single unit, private values setting that negative (or positive) externalities can lead to "non-strategic participation". Our multi-unit auction result is driven by the fear that free-riding will not be possible and, to a smaller extent, the desire to free-ride.

<sup>&</sup>lt;sup>17</sup>While we cannot say for sure that there are no other equilibria than those identified in Theorem 5, the general message that there will be short squeezing under the discriminatory auction is robust, since there are only mixed strategy equilibria.

<sup>&</sup>lt;sup>18</sup>When  $Q = Z < |y_{1,0}|$ , the probability of a short squeeze depends on  $\delta$ . If  $\delta = 1$ , the probability is zero because if the short buys all Z units, he avoids being squeezed on  $|y_{1,0}|$ . Hence, his average valuation is above  $R_h$ . If  $\delta = 0$ , the short and all longs with market power value all units at  $R_h$ . In equilibrium, they bid this and there will be a short squeeze for sure because of rationing. When  $Q = Z = |y_{1,0}|$ , there also will be a squeeze for sure because of rationing.

example, if there are several largest longs to begin with and none of them buys in the auction, or if all of them submit the same bids. The former could happen when the scope for free-riding is large; in this case, the competition between other longs and the short will sometimes be so intense that there is no benefit for any of the largest longs to become active. In contrast, when the scope for free-riding is small, all longs with market power have a strong incentive to bid aggressively so as to protect their market power.

# 5 Large Auctions: Implications and Predictions

In this section, we draw out empirical implications and predictions of our model. We focus on large uniform (Theorem 4) and discriminatory (Theorem 5) auctions with many longs where there is the potential for a short squeeze (Z > 0). Our findings are divided into two parts: bidder behavior and auction performance. It should be emphasized that the predictions of our model are, by design, pure implications of the loser's nightmare.

### 5.1 Bidder Behavior

We focus on measures of individual bidder behavior that have been used in the empirical literature; for example, number of bids, total quantity demanded, and quantity weighted mean price and variance [Gordy (1999) and Nyborg, Rydqvist, and Sundaresan (2001)]. Our findings delineate the differences in behavior between short and long bidders and are testable by someone possessing the appropriate data set.

When players use mixed strategies, as in our model in discriminatory auctions, what is most easily observed by an econometrician are the realizations of these mixed strategies. Our model delivers sharp predictions regarding these realizations, which can be tested by examining the bids submitted by short and long bidders in a given auction.

**Proposition 4** Given any realization of the equilibrium strategies of the players, the number of bids and the quantity weighted variance of the bids submitted by the short player tends to be larger than that of any long player, irrespective of the auction mechanism. Moreover, in uniform auctions, the average rate (or price) of the bids submitted by the short player is larger than that of any long player.

The first statement is a simple consequence of the observation that in the equilibria described in Theorems 4 and 5, the short submits two bids and the longs submit only one bid each. The second statement follows directly from the result that in uniform auctions, the short submits a bid at  $R_h$ , or higher, and (possibly) one at  $R_0$ , while the longs submit their bids at  $R_0$ . Note that in discriminatory auctions, it would be meaningless to compare the short and aggressive long's mean bids in a single auction since their bids are stochastic. However, since most longs are passive, we would expect to see the short place bids at higher mean rates than most longs. Furthermore, in discriminatory auctions: (i) shorts, but not longs, always demand some quantity at rates above the competitive rate, but (ii) when long players do so, they tend to demand more than shorts.

Next, we address the stochastic properties of equilibrium strategies in discriminatory auctions. The resulting propositions can be tested using a cross-section of auctions.

#### **Proposition 5** In discriminatory auctions, F(S) first order stochastically dominates $G(L; \delta)$ .

This expands the result for N = 2 illustrated in Figure 1. The intuition derives from the fact that the short values the first few units very highly compared with the average valuation the aggressive long places on all Q units – and recall that F(S) is the distribution used by the short for Z units, while  $G(L; \delta)$  is the distribution used by the long for Q units.

**Proposition 6** In discriminatory auctions,  $G(L; \delta)$  is an increasing function of the scope for free-riding,  $\delta$ .

This proposition shows that as the scope for free-riding increases, the active long becomes less aggressive in the sense of first order stochastic dominance. This can be understood by noting that it also means that the short becomes more aggressive in relative terms, which is intuitive since an increase in  $\delta$  translates into an increase in the cost to the short in the event of a squeeze. The reason that it is the long's, and not the short's, strategy that is affected by changes in the scope for free-riding is essentially that the short still needs Z units to cover. For the same reason, the short's strategy is the same as it were for N = 2, as revealed by a comparison between Theorems 2 and 5.

**Proposition 7** In discriminatory auctions, the expected quantity weighted mean bid for the short is greater or equal to that of the aggressive long (equal if and only if N = 2 or  $\delta = 0$ ).

The difference in the mean bids of the short and aggressive long is a consequence of differences in valuations. When there are no free-riders, the short's average valuation of the Q auctioned units is the same as that of the aggressive long, namely  $\bar{R}_{QZ}$ . However, when there are free-riders the short's average valuation is higher, since he then values the first few units above  $R_h$ . This leads the short to have a higher equilibrium mean bid than the aggressive long.

**Proposition 8** In discriminatory auctions, the expected quantity weighted mean bids for the short and aggressive long are strictly decreasing in auction size, Q.

This is a consequence of the price-quantity tradeoff faced by the aggressive long: when auction size grows, he has to buy more units to squeeze on the same number of units. Consequently the long's willingness to bid aggressively drops. In equilibrium, the short adjusts his bids downwards in response to the long's less aggressive bidding.

**Proposition 9** Suppose N = 2. In discriminatory auctions, the variance of the quantity weighted mean bid is larger for the long than for the short.

In the case of two players the short's variance is smaller primarily because the short always places a proportion of his bids at  $R_0$ . Additionally, the fact that the long has a mass point at  $R_0$  tends to augment the long's variance.

#### 5.2 Auction Performance

Here, we compare the performance of uniform and discriminatory auctions on three measures: revenue, the probability of a short squeeze, and (by implication) post-auction volatility. A clear message that has emerged from our analysis is that a short squeeze is more likely under discriminatory auctions than under uniform auctions, *ceteris paribus*. This appears to be a fundamental and robust result, arising from the different valuation schedules of the shorts and longs and the uniform pricing scheme, which allows shorts to bid very aggressively for a few units without having to worry about paying what they bid.

Initially, we focus on discriminatory auctions, since the outcome of uniform auctions is already summarized in Theorem 4. We start by looking deeper into the determinants of a short squeeze. Define  $\Theta$  to be the equilibrium probability of a short squeeze when the auction is discriminatory. For N = 2, Theorem 2 implies that  $\Theta = Z/2Q$ , which leads to the intuitively appealing results that  $\Theta$  is increasing in the market power of the long and decreasing in auction size. However, when N > 2, Theorem 5 implies that for  $\delta > 0$ ,

$$\Theta = \int_{[R_0, \bar{R}_{QZ}]} F(L) \, dG(L; \delta) = \frac{(Q - Z) \Big( -Z(|y_{1,0}| - Z)\delta + (Q\psi - Z^2) \ln \Big[ \frac{Q\psi - Z^2}{\psi(Q - Z)} \Big] \Big)}{Q(|y_{1,0}| - Z)^2 \delta^2}, \quad (18)$$

where  $\psi = Z(1 - \delta) + |y_{1,0}|\delta$ . For  $\delta = 0$ ,  $\Theta$  is the same as for N = 2.<sup>19</sup> This tells us that in the general case, the equilibrium probability of a short squeeze depends on the exogenous parameters in quite non-trivial ways. The change in  $\Theta$  with respect to Q can be decomposed into two effects:

$$\frac{\partial \Theta}{\partial Q} = \frac{\partial \int_{[R_0, \bar{R}_{QZ}]} F(L) dG(L; \delta)}{\partial Q} = \int_{R_0}^{\bar{R}_{QZ}} \frac{\partial F(L) G'(L; \delta)}{\partial Q} dL + G'(\bar{R}_{QZ}; \delta) \frac{\partial \bar{R}_{QZ}}{\partial Q}.$$
 (19)

The first term is a "direct" effect; it captures the effect an increase in Q has on  $F(\cdot)$  and  $G(\cdot; \delta)$ , keeping the upper bound of these distributions,  $\bar{R}_{QZ}$ , constant. The second term is an "indirect" effect; it captures the effect of the decrease in  $\bar{R}_{QZ}$  that results from an increase in Q. It can be shown that the "direct" effect is always positive and the "indirect" effect is always negative. A similar decomposition can be done for Z, but in this case the "direct" effect is negative and the "indirect" effect is positive.

When N = 2 or  $\delta = 0$ , the equilibrium probability of a short squeeze in discriminatory auctions is decreasing in auction size and increasing in Z. A similar result is reached in the following proposition for N > 2 and  $\delta > 0$ .

**Proposition 10** Suppose N > 2 and  $\delta > 0$ . (i) There is  $Q \ge Z$  such that for all Q > Q, the discriminatory auction equilibrium probability of a short squeeze,  $\Theta$ , is decreasing in the auction size, Q. Moreover  $\overline{Q}$  approaches Z as  $\delta$  goes to zero. (ii) There is  $\overline{Z}$  such that for all  $Z < \overline{Z}$ ,  $\Theta$  is increasing in Z. Moreover,  $\overline{Z}$  approaches Q as  $\delta$  goes to zero. (iii)  $\Theta$  is decreasing in the scope for free-riding,  $\delta$ . (iv)  $\Theta$  is decreasing in the absolute value of the short's initial holding,  $|y_{1,0}|$ . (v) In the limit as either Q or  $|y_{1,0}|$  becomes arbitrarily large,  $\Theta$  goes to zero.

<sup>&</sup>lt;sup>19</sup> Note also that for N > 2 and  $\delta > 0$ ,  $\lim_{\delta \to 0^+} \Theta = Z/2Q$ .

The first statement of the proposition shows that the indirect effect dominates for sufficiently large Q. This can be understood by recalling that  $\bar{R}_{QZ}$  is also the long's break-even value; it is the maximum that he is willing to pay for Q units. The valuation of the short, however, for the first Z units is unaffected by Q. So as  $\bar{R}_{QZ}$  falls, the valuation gap between the aggressive long and the short becomes increasingly large, hence the long's willingness to be aggressive falls relative to that of the short. The upshot is that the probability of a short squeeze falls when the auction becomes larger.

The second statement of the proposition has a similar intuition, but in this case  $R_{QZ}$  is increasing in Z. Hence, the equilibrium probability of a short squeeze tends to increase as the market power of the largest long increases. The third statement is a simple consequence of the result that  $G(L; \delta)$  is increasing in  $\delta$  while F(S) is not affected by  $\delta$ .

The fourth statement of the proposition can be understood by noting that for  $|y_{1,0}|$  there is no "indirect" effect since  $\bar{R}_{QZ}$  does not depend on  $|y_{1,0}|$ . The proposition shows that, *ceteris paribus*, as the short's position grows, the "direct" effect is negative. This is because more of the benefits to the longs from short squeezing will go to free-riders, thus leading the active long to bid less aggressively. It is important to keep in mind that this is a comparative statics result. In practice, it is likely that Z and  $|y_{1,0}|$  are positively correlated, but here Z is being kept constant.

Figure 2 depicts the probability of a short squeeze as a function of auction size. The figure shows the typical situation when N > 2 and  $\delta > 0$  that the probability of a short squeeze is initially increasing and then decreasing. Figure 3 depicts a similar, but opposite pattern for  $\Theta$  as a function of the largest long's market power. Figure 4 shows the effect on  $\Theta$  of changing the short's initial holding. Figures 2 and 4 also illustrate the fifth statement in Proposition 10. Intuitively, when auction size or the short's initial holding approaches infinity, the aggressive long and the short's bidding strategies "diverge"; that is, the short places almost all the mass of his bid for Z units in a neighborhood of  $\bar{R}_{QZ}$ , while the long's mass point at  $R_0$  approaches 1. Hence, the probability of a short squeeze approaches zero.

**Proposition 11** In the discriminatory auction equilibrium where long player n is aggressive, expected payoffs are: (i) The short:  $E[\pi_1] = y_{1,0}R_0 - (\bar{R}_{QZ} - R_0)Z$ . (ii) Aggressive long:  $E[\pi_n] = y_{2,0}R_0$ . (iii) Passive longs: for  $i \notin \{1, n\}$ ,  $E[\pi_i] = y_{i,0}[R_0 + \Theta\delta(R_h - R_0)]$ . (iv) Auctioneer (central bank):  $E[\pi_A] = QR_0 + Z(\bar{R}_{QZ} - R_0) - \sum_{i\geq 3} y_{i,0}(R_h - R_0)\delta\Theta$ .

This proposition shows that, in expectation, the short pays less than his valuation of  $R_h$  to cover the units he needs. Moreover, the aggressive long does not earn any rents from a squeeze – all his quasi-rents when a squeeze occurs go to the seller in the auction, through the rate the aggressive long ends up paying when there is no squeeze. Intuitively, this is because the short values the first few units higher than the long. Hence, competition from the short drives the rents of the active long to zero. This sheds light on why other longs are passive. From the short's perspective, the limited competition from the longs allows him to buy below his valuation. However, a comparison with Theorem 4 shows that the short does not do as well in discriminatory auctions as in uniform auctions. In contrast to the aggressive long, when there is some scope for free-riding, passive longs earn positive abnormal returns in expectation.

Proposition 11 also shows that the expected equilibrium payoffs to the short and aggressive long are unaffected by the scope for free-riding. This irrelevance result contrasts with the result on the equilibrium probability of a short squeeze above. It can be understood by an analogy to private value, single unit, second price auctions. In these auctions, the bidder with the highest valuation wins and pays the valuation of the second highest bidder. In our case, the short is the player with the higher valuation; he values the first Z units at  $R_h$ . The long values all Q units at an average of  $\bar{R}_{QZ} < R_h$ . Inspection of the short's expected equilibrium payoff reveals that it equals the value of his initial holding less the amount  $(\bar{R}_{QZ} - R_0)Z$ , which we can interpret as an excess payment over the "no squeeze" value of Z units. Essentially, the short is paying the valuation of the aggressive long. Put another way, the short's excess payment is the smallest amount that guarantees that he will not be squeezed. Since the aggressive long's valuation is not affected by the scope for free-riding, neither is the short's expected equilibrium payoff.

An implication of Proposition 11 is that the short's expected equilibrium payoff is increasing in the auction size, since  $\bar{R}_{QZ}$  is decreasing in Q. This is another contrast with the result on the probability of a short squeeze, which may exhibit a non-monotonic pattern as a function of Q (Figure 2). What happens when  $\Theta$  is increasing in Q is that this is offset by a lower average payment of the short in the event of no squeeze.

# **Proposition 12** Compared to discriminatory auctions, uniform auctions lead to (in expectation): lower auction revenue, fewer short squeezes, and lower post-auction volatility.

The result that auction revenue is higher in discriminatory auctions is interesting, not least because at a first glance it appears to be counterfactual. For example, empirical studies of US Treasury auctions show that markups (defined as the difference between auction and when-issued yields) are larger in discriminatory auctions than in uniform auctions [Nyborg and Sundaresan (1996), Malvey and Archibald (1998)]. This suggests a smaller revenue under discriminatory auctions. However, the problem with this inference is that when there is the potential for a short squeeze or when a squeeze is on, the when-issued yield will contain a "squeeze premium." In this case, the markup is not an accurate reflection of auction revenue (Nyborg and Sundaresan, 1996). Hence, revenue may be larger under discriminatory auctions even though the markup is larger, which is essentially what would be happening under our model. The larger revenue is a pure reflection of the increased likelihood of a short squeeze.

Short squeezing tends to lead to higher volatility in the secondary market, since some units change hands at  $R_h > R_0$ . Our result that the probability of short squeezing is higher after discriminatory auctions therefore means that volatility will tend to be higher too. This is consistent with the empirical evidence (Nyborg and Sundaresan, 1996). In the context of treasury auctions, for example, we would expect a short squeeze to manifest itself through the specialness of the on-the-run security. Thus, another testable prediction of our model is that specialness should explain more of volatility after discriminatory auctions than after uniform auctions.

Proposition 12 shows that the revenue advantage of discriminatory auctions comes at the cost of a higher incidence of short squeezes and consequently a more volatile secondary market. There may be no straight answer to the problem of choosing the auction format; it will depend on the aims and preferences of the auctioneer. Sellers who are more interested in maximizing revenue would do better with the discriminatory auction than with the uniform auction. Sellers who are more interested in minimizing market distortions would do better with the uniform auction.

The final two propositions examine the impact of auction size on short squeezing and revenue per unit in discriminatory auctions.

**Proposition 13** In discriminatory auctions, revenue per unit is monotonically decreasing in auction size.

This is quite intuitive, since the valuation gap between the short and the aggressive long is increasing in auction size.

**Proposition 14** As either  $\frac{Q}{Z} \to \infty$  or, for  $\delta > 0$ ,  $\frac{|y_{1,0}|}{Z} \to \infty$  the equilibrium outcome in the discriminatory auction approaches the outcome in the uniform auction.

This asymptotic equivalence is intuitive since when Q becomes very large with respect to Z, the cost of implementing a squeeze becomes very large relative to the benefit. Put another way, the break-even value for the long approaches the competitive rate.

# 6 Extensions

#### 6.1 Pre-Auction Market

Given our results that pre-auction allocations affect auction performance and payoffs, it is natural to ask how these positions are formed in the first place and what is the influence of the auction format. To make this extension of the model realistic, one would have to recognize that in many contexts the set of players in the forward market may be different than in the auction itself. For example, in the context of central bank repo auctions, commercial banks are subject to client withdrawals and deposits, and the auction is only open to banks. In US Treasury securities markets, pension funds and other institutions are often active in the when-issued market, but not in the auction. Non-bidders may trade in the pre-auction market perhaps because of hedging motives (Nyborg and Sundaresan, 1996). Since hedgers may have price sensitive demand, bidders' positions right before the auction may be a function of the forward prices that they have posted. Furthermore, different bidders may end up with different positions, perhaps because they have relations with different clients.

Intuition based on our model would suggest that prices in the pre-auction market will tend to be higher under discriminatory auctions than under uniform auctions, reflecting the larger likelihood of a short squeeze. As a result, it is possible that the open interest in the pre-auction market will be lower under discriminatory auctions. This could explain the finding that when-issued volume is considerably larger under uniform auctions (Nyborg and Sundaresan, 1996). A lower open interest may also serve to reduce the incidence of short squeezes and make the performance of discriminatory auctions appear more similar to uniform auctions. The cost would be reduced hedging under the discriminatory format.

#### 6.2 Multiple Shorts

Another issue is that pre-auction trading may result in multiple bidders with short positions. Supposing that there is at least one long bidder as well, we see that the shorts' valuation schedules are still downward sloping. Furthermore, the shorts need to buy Zunits in the auction to avoid being squeezed. Hence, equilibria of large uniform auctions are characterized by (some of) the shorts bidding  $R_h$  for Z units, with the remaining bids placed at  $R_0$ . As in Theorems 1 and 4, there are no short squeezes. For discriminatory auctions, the different valuation schedules of players still means that there are no pure strategy equilibria, implying that a squeeze will happen with positive probability. However, equilibrium strategies would be more complicated than in Theorem 5, because many shorts may choose to bid aggressively.

#### 6.3 Negative Initial Aggregate Position

Due to pre-auction demand by hedgers who do not participate in the auction, bidders in the auction may have a negative initial aggregate position  $(Q_0 < 0)$ . This would necessitate a change in the definition of market power, since in the absence of an auction a long player would have monopoly power over all his units (assuming hedgers would not sell back into the market). That is, Player n's date 0 market power would be  $z_{n,0} = \max[0, y_{n,0}]$ . Furthermore, shorts would have to buy a total of  $Q_0$  units in the auction to avoid being squeezed. Hence,  $Q_0$  would essentially take on the role of Z. So we would now define a large auction as one where  $Q > Q_0$ . The combination of  $Q_0 < 0$  and multiple shorts raises the interesting possibility that in a large auction, players that are short at date 0 will be able to implement a short squeeze at date 2 if they buy the entire auction. Hence, short players may have U-shaped valuation schedules. However, as before, the equilibrium probability of a short squeeze is zero under large uniform auctions and positive under large discriminatory auctions. As a consequence, expected auction revenue would be lower under uniform auctions than under discriminatory auctions.

### 6.4 Secondary Market

It also would be interesting to incorporate a more realistic model of the secondary market. Starting with Kyle (1984), a common feature of models of short squeezing is that a squeeze is only possible if some long player has a combined position in the forward and spot markets of the underlying asset of more than 100%. However, events surrounding the Salomon scandal demonstrate that squeezing can be implemented with a smaller position [Jegadeesh (1993), Jordan and Jordan (1996)]. Additionally, empirical evidence by Sundaresan (1994) as well as anecdotal evidence suggest that short squeezing happens from time to time in US Treasury auctions even when the 35% rule is observed. This suggests that the mechanics of a short squeeze are more subtle than what is currently modeled in the literature as well as in our paper. Capturing this would enrich our model and quite possibly lead to additional empirical predictions; however, we would expect the basic thrust of our conclusion regarding the difference between uniform and discriminatory auctions to remain valid.

# 7 Concluding Remarks

In this paper, we have studied the impact of a potential short squeeze in the secondary market on equilibrium bidding strategies in a multiple unit auction. What is important for our analysis is not the actual occurrence of a short squeeze, but the possibility of one. We have established closed form solutions for equilibrium strategies and outcomes and found that: (1) Players who have the opportunity to short squeeze, who have the potential to free-ride on a short squeeze, who are not affected by a short squeeze, or who risk being short squeezed, have different marginal valuations of the auctioned asset in an otherwise standard common value environment. (2) There is a trade-off between the two auction types that are most frequently used in practice, namely uniform and discriminatory auction. Specifically, discriminatory auctions lead to more price distortion and higher after-auction volatility, however, they also provide more revenue to the seller. (3) In discriminatory auctions, price distortion and the probability of a short squeeze depend on the size of the auction, market power of the largest longs, and the scope for free riding. (4) In uniform auctions, bidders with higher average bids spread their bids more. (5) In both types of auctions, short players tend to spread their bids more than long players. (6) The scope for free-riding does not affect the qualitative nature of equilibria in the two auction formats.

We have motivated our model with reference to the fact that many large and important multi-unit auctions in financial and commodity markets are embedded in larger structures, where bidders often enter the auction with short positions and must cover in the auction or, failing that, in the after-market where they can be squeezed. More generally, our model is also applicable to secondary market situations where a remedy against a short squeeze is sought.<sup>20</sup> A recent squeeze took place in the gold markets at the end of February 2001:

[Gold lease] rates have been squeezed considerably by the requirements of borrowers versus a lack of lenders. Yesterday afternoon, there was substantial lending from one of the central banks...(*Dow Jones Newswires, February 2001*)

The jump in the one month gold lease rate from its normal level of less than 1% to above 4.5% in a matter of days is testimony to this particular short squeeze. The fact that the 12 month rate was at most around 2% during this time is also consistent with a short squeeze. One may attempt to alleviate a short squeeze by releasing additional stock to the market, as indeed seems to have happened in our gold squeeze example. However, this is likely to be a blunt instrument unless one can find a way of actually selling to the

<sup>&</sup>lt;sup>20</sup>Well known short squeezes in commodity markets include the Hunt brothers' silver squeeze in the late 1970's and the Sumitomo copper squeeze in the 1990's. Other examples in various markets include: 30 year US Treasury bonds (*The Economist*, 12 February 2000), the German government bond futures market (*eFinancial News*, 26 March 2001) ["Deutsche Bank makes Euro 50m from Bobl squeeze."], ABN AMRO (on the Lander 5 bond), (*Euroweek*, 15 May 1998), Japanese government bond futures market (*Barron's*, 28 October 1996) ["A giant short squeeze involving Japanese government bonds futures last month...apparently prompted actions by Japanese regulators. The squeeze...was the largest in the history of global financial futures markets..."], NYMEX gas futures (*Futures*, February 1996), Apex Oil (*Forbes*, 24 May 1982). Backwardation in commodity futures is also often attributed to short squeezes or potential short squeezes. In the equity markets, small cap/small float stocks seem to be particularly vulnerable to short squeezes. Examples include: MicroStrategy (*Washington Business Journal*, 16 June 2000), Solv-Ex (*Business Week*, 5 August 1996), Presstek (*Forbes*, 17 June 1996).

shorts rather than the longs. In practice, trying to locate and verify who is short may be difficult, costly, and time consuming. However, our analysis suggests that the uniform auction may be an effective mechanism that achieves the objective of selling to the shorts, and the auction can be organized quickly and relatively cheaply.

There are numerous ways to extend our model. We have already discussed some of them in the previous section. Other issues that would be interesting to pursue in the context of our framework include: How does private information about allocations affect the results? What is the impact of private information about the competitive and "squeeze" rates? Is there a relation between the winner's curse and short squeezing? For example, private information may give positive rents to a short squeezer. Supply uncertainty would also be interesting to examine. For instance, in US treasury auctions, non-competitive demand creates supply uncertainty for competitive bidders. We believe that the relative simplicity of our model would make it a useful building block in examining these and other issues. Finally, the richness of the model's predictions along with its relative simple structure makes our model potentially useful in experimental settings.

# 8 Appendix

#### Proof of Theorem 1

Suppose that Player 1 submits  $b_1^u$ . If Player 2 does nothing or submits  $b_2^u$ , his payoff will be  $R_0y_{2,0}$ . The only possible way for Player 2 to improve on this is to implement a squeeze. To do this, Player 2 must bid above  $R_h$  for at least  $Q' \ge Q - Z + 1$  units. If he does so, his payoff is

$$\pi_2 = R_0 y_{2,0} + (R_h - R_0)(Z - (Q - Q')) - (a_{2,1} - R_0)Q' < R_0 y_{2,0} + (R_h - R_0)(Z - Q) < R_0 y_{2,0},$$

using (8), the fact that  $z_{2,2} = Z - (Q - y_{2,1}) = Z - (Q - Q')$ , and  $a_{2,1} < R_h$ . Hence, Player 2 cannot do better than submitting  $\mathbf{b}_2^u$  in response to  $\mathbf{b}_1^u$ . Since  $\mathbf{b}_1^u$  is a best response to  $\mathbf{b}_2^u$ , these strategies form an equilibrium. To show that any other equilibrium will give the same outcome as  $(\mathbf{b}_1^u, \mathbf{b}_2^u)$ , we observe first that it is always a best response for Player 1 to submit  $\mathbf{b}_1^u$  as long as Player 2 does not demand Q units at  $R_h$  or higher. Second, since (as observed in the text) the most Player 2 would be willing to pay for Q units is  $\overline{R}_{QZ} < R_h$ , it follows that any other equilibrium is a trivial variation of  $(\mathbf{b}_1^u, \mathbf{b}_2^u)$ . Finally, since in equilibrium Player 1 wins Z units, there is no short-squeezing in the secondary market.

#### Proof of Lemma 1

Note first that under pure strategies, the auction outcome is known with certainty. Furthermore, it is not consistent with equilibrium for the short to pay more than  $R_0$  for more than Z units and therefore for the long to demand more than  $\bar{R}_{QZ}$  for any unit.

Suppose first that there is a pure strategy equilibrium when  $z_{2,2} = 0$ . It would be inconsistent with this for the long to pay more than  $R_0$  for any unit in the auction, and therefore, for the short to pay more than  $R_0 + \epsilon$  for any unit, where  $\epsilon \ge 0$  is arbitrarily small. Hence, the highest bid is at most  $R_0 + \epsilon$  and is made by the short. But then the long could do better by submitting a bid for Q units at  $R_0 + \epsilon + \epsilon_1$ , where  $\epsilon_1 > 0$  is arbitrarily small. This is a contradiction.

Suppose next that there is an equilibrium in pure strategies when  $z_{2,2} > 0$ . This means that the long buys more than Q - Z units in the auction. Define  $r_z$  to be the rate at which the long

wins his Q - Z + 1th unit.<sup>21</sup> In equilibrium, we must have  $r_z \leq \bar{R}_{QZ}$ . But then, since the short values his first Z units at  $R_h > \bar{R}_{QZ}$ , the short can improve his payoff by submitting a bid for 1 unit at  $r_z + \epsilon$ , where  $\epsilon > 0$  is arbitrarily small. This is a contradiction.

#### Proof of Theorem 2

Here we complete the proof started in the text. Under the proposed strategies, expected payoffs are  $E[\pi_S] = -Z\bar{R}_{QZ}$  and  $E[\pi_L] = y_{2,0}R_0$ . We first examine whether the short can improve on this by splitting his bid for Z units. Suppose the short splits this into Z bids,  $\{(S_k, 1)\}_{k=1}^Z$ , where  $S_k \geq S_{k+1}$  and  $S_k \in [R_0, \bar{R}_{QZ}]^{22}$  He also submits a bid for Q-Z units at  $R_0$  as before. Then<sup>23</sup>

$$E[\pi_S] = \sum_{k=1}^{Z} G(S_k)(R_h - S_k) - ZR_h = -Z\bar{R}_{QZ}.$$
(20)

Hence, no matter how the short splits his bid for Z units, his payoff remains  $-Z\bar{R}_{QZ}$ .

Next, we examine whether the long might improve his payoff by splitting his bid. So suppose the long splits this into Q bids,  $\{(L_k, 1)\}_{k=1}^Q$ , where  $L_k \ge L_{k+1}$  and  $L_k \in [R_0, \bar{R}_{QZ}]$ . Then<sup>24</sup>

$$E[\pi_L] = R_0 y_{2,0} + \frac{1}{Z} \Big( (Q - Z) \sum_{k=Q-Z+1}^Q L_k - Z \sum_{k=1}^{Q-Z} L_k \Big).$$
(21)

Notice that the second and the third parts of this expression have the same number of terms ((Q-Z)Z). Since  $L_k \geq L_{k+1}$ , it follows that the last two terms sum to something which is at most zero. Hence, no matter how the long splits his bid for Q units, he cannot improve on his expected payoff of  $R_0 y_{2,0}$ . This establishes the theorem. 

#### Proof of Theorem 3

If a player bids for less than Q units, the other player can pick up some units at a rate below  $R_0$ , which is not consistent with equilibrium. Hence, in a single bid equilibrium, at least one of  $b_1$  or  $b_2$  has form:  $(\tilde{r}, Q)$ . But  $b_1$  cannot take this form if  $\Pr(\tilde{r} > R_0) > 0$ , since the short values only the first Z units above  $R_0$ . Additionally, Lemma 1 shows that  $b_1 = \{(R_0, Q)\}$  is not consistent with equilibrium. Therefore, in a single bid equilibrium,  $b_2 = \{(\tilde{r}, Q)\}$ . Hence, by the proof of Theorem 2, it only remains to show that the support of  $F(\cdot)$  and  $G(\cdot)$  is  $[R_0, \bar{R}_{QZ}]$  and that  $F(\cdot)$ and  $G(\cdot)$  do not have coinciding mass points. Clearly, the lower bound of the support is  $R_0$ . But suppose that the upper bound of the support is  $R < R_{QZ}$ . Solving for F(S) and G(L) as before (Theorem 2), we find that both F(S) and G(L) have mass points at  $R_0$ . But then, for some  $\epsilon > 0$ , the long could increase his payoff by placing his mass point at  $R_0 + \epsilon$ . Similarly, if  $F(\cdot)$  and  $G(\cdot)$  have coinciding mass points above  $R_0$ , at least one of the players could improve his payoff by moving the mass point up slightly. Since F(L) and G(S) are the unique solutions for particular boundaries when there are no coinciding mass points, this concludes the proof.  $\Box$ 

<sup>23</sup>We start the calculation of (20) by noting that:  $E[\pi_S] = G(S_Z)(\sum_{k=1}^Z -S_k) + (G(S_{Z-1}) - S_Z)$ 

 $\begin{aligned} & G(S_Z))((\sum_{k=1}^{Z-1} - S_k) - R_h) + \ldots + (G(S_2) - G(S_1))(-S_1 - (Z-1)R_h) + (1 - G(S_1))(-ZR_h). \\ & (2^4 \text{We start the calculation of (21) by noting that: } E[\pi_L] = R_0 y_{2,0} + (Q-Z)R_0 + F(L_Q)(ZR_h - \sum_{k=1}^{Q} L_k) + (F(L_{Q-1}) - F(L_Q))((Z-1)R_h - \sum_{k=1}^{Q-1} L_k) + (F(L_{Q-Z}) - F(L_{Q-Z+1}))(-\sum_{k=1}^{Q-Z} L_k) + \ldots + (1 - F(L_1))(-\sum_{k=1}^{Q-Z} L_k) = R_0 y_{2,0} + \sum_{k=Q-Z+1}^{Q} F(L_k)(R_h - L_k) - \sum_{k=1}^{Q-Z} L_k. \end{aligned}$ 

<sup>&</sup>lt;sup>21</sup>If rationing at the stop-out rate means that the long only wins  $Q - Z + \theta$  units, where  $\theta < 1, r_z$  is the rate at which the long wins his last  $\theta$  units.

<sup>&</sup>lt;sup>22</sup>This specification is completely general, since a bid at  $R_0$  gives the same payoff as no bid at all. The specification allows for the short submitting any number  $Z' \leq Z$  of bids at any prices above  $R_0$ . As established above, the short cannot improve his payoff by demanding more than Z units above  $R_0$ .

#### **Proof of Proposition 2**

Consider the case  $\delta = 1$ , since it will maximize the potential benefits from entry. Under the proposed strategies, to stop the short from covering in the auction, new entrants must get at least Q - Z + 1 units. Suppose new entrants buy  $Q' \ge Q - Z + 1$  units. Under the proposed strategies, the short gets max[0, Q - Q'] units. Hence,  $\forall n \ z_{n,2} = 0$ . Thus, new entrants would earn negative profits since they would have to pay more than  $R_0$  on the Q' units obtained in the auction, so they would be better off not participating at all (or placing bids at  $R_0$ ). **Proof of Proposition 3** 

Consider the case  $\delta = 1$ , since it will maximize the potential benefits from entry. If any player with zero initial endowment bids for 1 unit at  $l > R_0$ , his expected payoff is

$$E[\pi_E] = (R_0 - l) \int_{[R_0, l]} (1 - F(L)) dG(L) + (R_h - l) \int_{[R_0, l]} F(L) dG(L).$$

Note that  $E[\pi_E|l=R_0]=0$  and

$$\frac{\partial E[\pi_E]}{\partial l} = C \left[ Q(l - R_0) - Z(R_0 - R_h) \right],$$

where  $C = \frac{(R_0 - R_h)^2 (Q - Z)}{QZ(R_h - L)^3}$  and so  $\frac{\partial E[\pi_E]}{\partial l} < 0$  for  $l < \bar{R}_{QZ}$ . Hence,  $E[\pi_E] < 0$  for all  $l > R_0$ . This expands trivially if the potential entrant bids for more than 1 unit up to Z - 1 units. There is no benefit from bidding for Z or more units, since if the potential new entrant were awarded this, there would be no short squeezing.

#### **Proof of Theorem 4**

The proof is virtually the same as for the case N = 2 (see the proof of Theorem 1).  $\Box$ **Proof of Theorem 5** 

We will prove the Theorem for  $q_{1,2}^* = Q - Z$ . The case that  $q_{1,2}^* < Q - Z$  follows immediately. Parts (i) and (ii): Let F(S) and  $G(L; \delta)$  denote the random parts of the short's and aggressive long's strategies, respectively. We wish to show that they are given by (16) and (17), respectively. Using the same procedure as in the proof of Theorem 2, we see that F(S) is given by (16). Furthermore,  $G(L; \delta)$  is derived by setting the integrand of the following integral equal to a constant:

$$E[\pi_1] = \int_{[R_0, \bar{R}_{QZ}]} \left\{ G(S; \delta) (-ZS - (|y_{1,0}| - Z)R_0) + (1 - G(S; \delta)) \left( -ZR_h - (|y_{1,0}| - Z)(R_0 + \delta(R_h - R_0)) \right) \right\} dF(S),$$
(22)

since under the proposed strategies X = 1 in the event of a squeeze [see (7)], which implies that  $G(L; \delta)$  is given by (17). The proof that the long will not deviate is the same as in Theorem 2, since the short's strategy is the same. Suppose now that the short deviates from the described strategy and splits his bid for Z units into Z bids (assuming  $Z \ge 2$ ),  $\{(S_k, 1)\}_{k=1}^Z$ , where  $S_k \ge S_{k+1}$  and  $S_k \in [R_0, \bar{R}_{QZ}]$ . Then, proceeding as in Theorem 2, we get

$$E[\pi_1] = \sum_{k=1}^{Z} G(S_k; \delta)(R_h - S_k) - ZR_h - (1 - G(S_Z; \delta))(|y_{1,0}| - Z)\delta(R_h - R_0) - (|y_{1,0}| - Z)R_0.$$
(23)

The expression above is strictly increasing in  $S_Z$ . Hence, the short will set  $S_Z = S_{Z-1}$ . Proceeding in the same vein, we get  $S_k = S_{k-1}$ ,  $k = 2 \dots Z - 1$ . Thus, the short maximizes his payoff by bidding  $S_1$  for all Z units and therefore cannot improve his payoff by deviating from  $b_1^*$ . Part (*iii*): For i = K + 2, ..., N,  $z_{i,0} = 0$ , Part (*iii*) can be established along the same lines as the no market-entry result in Proposition 3. Next, suppose that i = 2, ..., K + 1 and  $i \neq n$ . Suppose first that  $Z \ge 2$ . Then if a player bids for one unit at rate l,  $R_0 \le l \le \bar{R}_{QZ}$ , the change in his expected payoff relative to what he gets by being passive (excess payoff) is

$$\Delta \pi_i = (R_0 - l) \int_{[R_0, l]} (1 - F(L)) dG(L; \delta) + \left[ \delta(R_h - R_0) + (R_0 - l) \right] \int_{[R_0, l]} F(L) dG(L; \delta).$$

Thus  $\frac{\partial \Delta \pi_i}{\partial l} = \frac{\partial G(l;\delta)}{\partial l} [R_0 - l + \delta(R_h - R_0)F(l)] - G(l;\delta) = C[R_0(Z - Q)\delta + R_h(Z(\delta - 1) - |y_{1,0}|\delta) + L(Z + (Q + |y_{1,0}|)\delta - 2Z\delta)]$ , where C > 0 (see footnote 26). Since  $\frac{\partial^2 \Delta \pi_i}{\partial l^2} > 0$  and  $\frac{\partial \Delta \pi_i}{\partial l}|_{l=\bar{R}_{QZ}} = C \times \frac{(R_h - R_0)(Q - Z)(Z(2\delta - 1) - |y_{1,0}|\delta)}{Q} < 0$ , it follows that  $\frac{\partial \Delta \pi_i}{\partial l} < 0$ . Therefore, since  $\Delta \pi_i|_{l=R_0} = 0$ , the player is worse off by bidding for one unit at  $l > R_0$ . This expands trivially to the case of bidding for up to Z - 1 units. Suppose now that  $Z \ge 1$  and consider bids by Player *i* for  $Q' \ge Z$  units. Note first that it is not profitable to bid for  $Q' \le Q - z_{i,0}$  units, since there will be no short squeezing at all when l > L. Suppose therefore that Player *i* bids for  $Q' \in [Q - (z_{i,0} - 1), Q]$  units. Then his expected excess payoff is at most:<sup>25</sup>

$$\Delta \pi_{i} = [(Q' - Q + z_{i,0})R_{h} + (Q - z_{i,0})R_{0} + y_{i,0}R_{0} - Q'l - y_{i,0}R_{0} - y_{i,0}\delta(R_{h} - R_{0})]\Pr\{l > L > S\} + [(Q' - Q + z_{i,0})R_{h} + (Q - z_{i,0})R_{0} - Q'l + y_{i,0}R_{0} - y_{i,0}R_{0}]\Pr\{l > S > L\} + [(Q - Z)(R_{0} - l) + y_{i,0}R_{0} - y_{i,0}R_{0}]\Pr\{S > l > L\} = [(Q' - Q + z_{i,0})R_{h} + (Q - z_{i,0})R_{0} - Q'l]F(l)G(l;\delta)) + (Q - Z)(R_{0} - l)G(l;\delta))(1 - F(l)) - y_{i,0}\delta(R_{h} - R_{0})\int_{[R_{0},l]}F(L)dG(L;\delta)$$
(24)

The last term is always nonpositive, and the sum of the first two terms is less than  $\gamma(z_{i,0} - Z)$ , where  $\gamma > 0$  (since  $Q' \leq Q$ ).<sup>26</sup> Therefore Player *i* is worse off, since  $Z > z_{i,0}$ . Along the same lines as in Part (*ii*), Player *i* cannot earn a positive excess payoff by splitting his bids.

Final part: That any player without market power cannot benefit from bidding aggressively is a simple consequence of Proposition 3. Suppose therefore that some Player  $i \in M + 2, \ldots, K + 1$ bids aggressively and that all other longs are passive. As established in Theorem 3, it must be the aggressive long who submits only one bid. By the same procedure as above, we find that F(S) and  $G(L; \delta)$  are given by (16) and (17), respectively, but where Z should be replaced by  $z_{i,0}$  (this should also be done in the expression for the upper bound of the support). Given these strategies, suppose now that Player 2 (who has  $Z > z_{i,0}$  units of market power) deviates by bidding for Q units at a rate  $l > R_0$ . Then Player 2's expected excess payoff is:

$$\Delta \pi_{2} = [ZR_{h} + (Q - Z)R_{0} - Ql]F(l)G(l;\delta) - y_{2,0}\delta(R_{h} - R_{0})\int_{[R_{0},l]}F(L)dG(L;\delta) + [(Z - z_{i,0})R_{h} + (Q - Z)R_{0} - (Q - z_{i,0})l]G(l;\delta)(1 - F(l)) > [ZR_{h} + (Q - Z)R_{0} - Ql]F(l)G(l;\delta) + (Q - Z)(R_{0} - l)G(l;\delta)(1 - F(l)) - (|y_{1,0}| - z_{i,0})(R_{h} - R_{0})F(l)(G(l;\delta) - G(R_{0};\delta)),$$

$$(25)$$

since (a)  $(Q-Z)(R_0-l) < (Z-z_{i,0})R_h + (Q-Z)R_0 - (Q-z_{i,0})l$ ; (b)  $y_{2,0} \leq |y_{1,0}| - z_{i,0}$ ; (c)  $\int_{[R_0,l]} F(L)dG(L;\delta) < F(l)[G(l;\delta) - G(R_0;\delta)]$  (the latter is true since F(L) is monotonically increasing); and (d)  $\delta \leq 1$ . Thus, by direct substitution of F(L) and  $G(L;\delta)$  in the right hand side of (25) we get

$$\frac{(l-R_0)(R_h-R_0)(Q(z_{i,0}(\delta-1)-|y_{1,0}|\delta)+z_{i,0}^2)}{Q(R_h-L)(Lz_{i,0}+R_0(|y_{1,0}|-z_{i,0})\delta+R_h(z_{i,0}(\delta-1)-|y_{1,0}|\delta))} \times \left[\frac{(l-R_0)(Q-z_{i,0})(|y_{1,0}|-z_{i,0})}{|y_{1,0}|} + \frac{(L-R_h)Q(Z-z_{i,0})}{z_{i,0}}\right]^{25} \text{It is less in the event that } l > L > S \text{ and } X = 2.$$

$${}^{26}\gamma = \frac{(l-R_0)(R_h-R_0)^2(Q-Z)(Q(|y_{1,0}|\delta-Z(\delta-1))-Z^2)}{Q(R_h-l)Z(LZ+R_0(|y_{1,0}|-Z)\delta+R_h(Z(\delta-1)-|y_{1,0}|\delta))} > 0. \ C = \frac{(R_h-R_0)^2(Q(|y_{1,0}|\delta-Z(\delta-1))-Z^2)}{Q(R_h-L)(LZ+R_0(|y_{1,0}|-Z)\delta+R_h(Z(\delta-1)-|y_{1,0}|\delta))^2} > 0.$$

The first factor is negative  $\forall l$ . Moreover,  $\exists \epsilon > 0$ , small enough, so when  $l = R_0 + \epsilon$ , the second factor is always negative (since the first fraction is always positive and equal to 0 at  $l = R_0$ , and the second fraction is always negative). Hence, there is no equilibrium in which any one of the non-largest longs is aggressive with all other longs being passive.  $\square$ 

#### **Proof of Theorem 6**

(i) Follows directly from Q < Z. (ii) That there is a squeeze for sure implies: (1) Player 1 values each of the Q units in the auction at  $R_h$ ; and (2) for every auctioned unit and for any allocation of the Q-1 other units, at least one of the long players value the unit at  $R_h$ . Competition in the auction then implies that every unit will be sold for  $R_h$ . 

#### **Proof of Proposition 4**

The proposition follows from inspection of the strategies described in Theorems 4 and 5. **Proof of Proposition 5** 

First order stochastic dominance is obvious from inspection of F(S) and  $G(L; \delta)$ . **Proof of Proposition 6** 

The proposition follows from inspection of the derivative of  $G(L; \delta)$  with respect to  $\delta$ . **Proof of Proposition 7** 

We will consider the case  $q_{1,2}^* = Q - Z$ . It is obvious that this will minimize the expected mean bid of the short player. Let  $\mathbf{m}_i$  stand for the quantity-weighted mean bid of the bidder *i*. The expected mean bid for the short and aggressive long are, respectively,

$$E[\mathbf{m}_{\mathbf{S}}] = \frac{1}{Q} \int_{[R_0, \bar{R}_{QZ}]} (SZ + (Q - Z)R_0) dF(S) = R_0 + (R_h - R_0) \Big( Z - (Q - Z) \ln \frac{Q}{Q - Z} \Big) / Q$$

and

$$E[\mathbf{m}_{\mathbf{L}}] = G(R_0; \delta)R_0 + \int_{(R_0, \bar{R}_{QZ}]} LdG(L; \delta) = \left(Z((Q-Z)R_0 + R_hZ) - (R_h - R_0)(Q\psi - Z^2)\ln\frac{Q\psi}{Q\psi - Z^2}\right) / (QZ),$$

where  $\psi = Z(1-\delta) + |y_{1,0}|\delta$ . Direct calculation shows that

$$\lambda \equiv E[\mathbf{m}_{\mathbf{S}}] - E[\mathbf{m}_{\mathbf{L}}] = \frac{R_0 - R_h}{QZ} \Big[ (Q - Z) Z \ln(\frac{Q}{Q - Z}) - (Q\psi - Z^2) \ln(\frac{Q\psi}{Q\psi - Z^2}) \Big].$$

For N = 2, Theorem 2 shows that  $q_{1,2}^* = Q - Z$ . Since in this case  $|y_{1,0}| = Z$ , expectations are equal. When N > 2 and  $\delta = 0$ ,  $\psi = Z$  and so  $\lambda = 0$ . For N > 2 and  $\delta > 0$   $\frac{\partial \lambda}{\partial |y_{1,0}|} =$  $\frac{Z^2}{\psi} - Q \ln(\frac{Q\psi}{Q\psi - Z^2}) < 0. \text{ The inequality follows from the fact that } \frac{\partial^2 \lambda}{\partial \psi^2} = \frac{Z^4}{\psi - Z^2} > 0, \lim_{\psi \to \infty} \frac{\partial \lambda}{\partial \psi} = 0,$ and  $\exists \psi$  such that  $\frac{\partial^2 \lambda}{\partial \psi} < 0$ .

### Proof of Proposition 8

Using the results from proof of proposition 7, we have  $\frac{\partial E[\mathbf{m}_{\mathbf{S}}]}{\partial Q} = \frac{(R_0 - R_h)Z \ln \frac{Q}{Q-Z}}{Q^2} < 0$  and  $\frac{\partial E[\mathbf{m}_{\mathbf{L}}]}{\partial Q} =$  $\frac{\frac{(R_0-R_h)Z\ln\frac{Q\psi}{Q\psi-Z^2}}{Q^2}<0.$  **Proof of Proposition 9** 

If N = 2,  $q_{1,2}^* = Q - Z$  (see Theorem 2). In this case,

$$\sigma^{2}[\mathbf{m}_{\mathbf{S}}] - \sigma^{2}[\mathbf{m}_{\mathbf{L}}] = \frac{1}{Q} \Big( \int_{[R_{0}, \bar{R}_{Qz_{n}}]} (S + \frac{Q - Z}{Q} R_{0} - E[\mathbf{m}_{\mathbf{S}}])^{2} dF(S) - \int_{(R_{0}, \bar{R}_{Qz_{n}}]} (L - E[\mathbf{m}_{\mathbf{L}}])^{2} dG(L; \delta) - \frac{Q - Z}{Q} (R_{0} - E[\mathbf{m}_{\mathbf{L}}])^{2} \Big) = \frac{1}{Q^{2}} (R_{0} - R_{h})^{2} (Q - Z) \varphi,$$

where  $\varphi \equiv Z(Z - 2Q) + 2Q(Q - Z) \ln \frac{Q}{Q - Z}$ . To establish the proposition, it suffices to show that  $\varphi < 0$ . Now,  $\frac{\partial \varphi(Q,Z)}{\partial Q} = 2(Z - Q \ln \frac{Q}{Q - Z})$  and  $\frac{\partial^2 \varphi(Q,Z)}{\partial Z^2} = \frac{2Z}{Z - Q} < 0$ . Since  $\frac{\partial \varphi(Q,Z)}{\partial Q}|_{Z=0} = 0$ , this implies that  $\frac{\partial \varphi(Q,Z)}{\partial Q} < 0$ . Hence, since  $\varphi(Q,Z)|_{Z=0} = 0$ , we have  $\varphi(Q,Z) < 0$ . **Proof of Proposition 10** 

Part (i): Using (18), we obtain

$$\frac{\partial\Theta}{\partial Q} = \frac{-(|y_{1,0}| - Z)Z(Q + Z)\delta + (Q^2\psi - Z^3)\ln\left[\frac{Q\psi - Z^2}{\psi(Q - Z)}\right]}{Q^2(|y_{1,0}| - Z)^2\delta^2}.$$
(26)

Since  $\ln(1+x) \leq x - \frac{x^2}{2} + \frac{x^3}{3}$ , we may get an upper bound on (26) by substituting the above approximation for the "ln" term, where  $x = \frac{Z(\psi-Z)}{\psi(Q-Z)}$ :

$$\frac{\partial\Theta}{\partial Q} \le \frac{Z^2(|y_{1,0}| - Z)^2 \delta^2}{6(Q - Z)^3 \psi^3 Q^2(|y_{1,0}| - Z)^2 \delta^2} \times [a_1 Q^3 + o(Q^2)],\tag{27}$$

where  $a_1 = -(1 - \delta)Z(Z + 2|y_{1,0}|\delta) - Z^2\delta^2 < 0$ . Clearly  $\exists \bar{Q}$  such that  $\forall Q > \bar{Q}$  this expression is negative, which is sufficient to establish Part (*i*). The limiting result follows from the result that for N > 2 and  $\delta > 0$ ,  $\lim_{\delta \to 0^+} \Theta = Z/2Q$ . The proof of part (*ii*) goes along the same lines. Part (*iii*): Proceeding as in Part (*i*), we get an upper bound on  $\frac{\partial \Theta}{\partial \delta}$  using  $\ln(1 + x) < x$ :

$$\frac{\partial \Theta}{\partial \delta} \le -\frac{Z^2}{Q\delta\psi} < 0, \tag{28}$$

Part (iv):  $\frac{\partial \Theta}{\partial |y_{1,0}|} = \int_{R_0}^{\bar{R}_{QZ}} \frac{\partial F(L)G'(L;\delta)}{\partial |y_{1,0}|} dL < 0$ , since the integrand is less than zero for any  $L \in (R_0, \bar{R}_{QZ}]$ . Part (v): Use L'Hospital's rule.

#### **Proof of Proposition 11**

The expected equilibrium payoffs to the short and the aggressive long can be calculated as in the proof of Theorem 2, which deals with N = 2, using (22) and (13), respectively. Consider next the "passive" Player  $i, i \notin \{1, n\}$ . Player i gets a payoff of  $y_{i,0}(R_0 + \delta(R_h - R_0))$  in the event of a short squeeze (probability  $\Theta$ ) and  $y_{i,0}R_0$  in the event of no squeeze. Hence,  $E[\pi_i]$  is as stated in the proposition. Finally, note that the total payoff always equals  $(Q + \sum_{i=1}^{N} y_{i,0})R_0$ . Therefore the expected interest earnings of the auctioneer can be calculated by subtracting the payoffs of all other players from this expression and using the definition of Z.

#### **Proof of Proposition 12**

Follows from a direct comparison of Proposition 11 and Theorem 4.

#### **Proof of Proposition 13**

Using  $Z = |y_{1,0}| - \sum_{i>3} y_{i,0}$  and (18) in the expression for  $E[\pi_A]$  in Proposition 11, we get

$$\frac{\partial \frac{E[\pi_A]}{Q}}{\partial Q} = \frac{(R_h - R_0)Z(2Z^2 - Q(\psi + Z))}{Q^3(|y_{1,0}| - Z)\delta} \times \ln\left[\frac{Q\psi - Z^2}{\psi(Q - Z)}\right].$$

Since Q > Z and  $\psi \ge Z$ , the first factor is negative and the second is positive. The Proposition follows.

#### **Proof of Proposition 14**

As  $\frac{Q}{Z} \to \infty$ ,  $\overline{R}_{QZ} \to R_0$  and therefore  $E[\pi_1] \to y_{1,0}R_0$  and both F(S) and  $G(S;\delta)$  go to  $R_0$ . It follows that  $\Theta$  approaches zero, therefore  $E[\pi_A] \to QR_0$  and  $\forall i E[\pi_i] \to y_{i,0}R_0$ . For  $\delta > 0$  the case that  $\frac{|y_{1,0}|}{Z} \to \infty$  follows along the same lines.  $\Box$ 

# 9 References

- Allen, B. and M. Hellwig, 1986, "Bertrand-Edgeworth Oligopoly in Large Markets," Review of Economic Studies 53, 175-204.
- 2. Back, K., and J. F. Zender, 1993, "Auctions of Divisible Goods: On the Rationale for the Treasury Experiment" *Review of Financial Studies*, 6, 733-76.
- Bergstrom, T., L. Blume, and H. Varian, 1986, "On the Provision of Public Goods," Journal of Public Economics 29, 25-49.
- 4. Bikhchandani, S., and Chi-Fu Huang, 1989, "Auctions With Resale Markets: A Model of Treasury Bill Auctions," *Review of Financial Studies* 2, 311-40.
- Bikhchandani, S., and Chi-Fu Huang, 1993, "The Economics of Treasury Securities Markets," *Journal of Economic Perspectives* 7, 117-34.
- Chatterjea, A., and R. A. Jarrow, 1998, "Market Manipulation, Price Bubbles, and a Model of the U.S. Treasury Securities Auction" Journal of Financial and Quantitative Analysis, 33, 255-89.
- Cooper, D. J., and R. G. Donaldson, 1998, "A Strategic Analysis of Corners and Squeezes" Journal of Financial and Quantitative Analysis, 33, 117-37.
- Dunn, K. B., and C.S. Spatt, 1984, "A Strategic Analysis of Sinking Fund Bonds", Journal of Financial Economics, 13, 399-423.
- 9. Gordy, M.B., 1999, "Hedging Winner's Curse with Multiple Bids: Evidence from the Portuguese Treasury Bill Auction," *Review of Economics and Statistics* 81, 448-465.
- Hamilton, J. D., 1996, "The Daily Market for Federal Funds," Journal of Political Economy 104, 26-56.
- 11. Hartmann, P., Manna, M., and A. Manzanares, 2001, "The Microstructure of the Euro Money Market", *Journal of International Money and Finance* 20, 897-952.
- Jarrow, R. A., 1992, "Market Manipulation, Bubbles, Corners, and Short Squeezes", Journal of Financial and Quantitative Analysis 27, 311-336.
- Jarrow, R. A., 1994, "Derivative Security Markets, Market Manipulation, and Option Pricing Theory", Journal of Financial and Quantitative Analysis 29, 241-261.
- Jegadeesh, N., 1993, "Treasury Auction Bids and the Salomon Squeeze", Journal of Finance 48, 1403-1419.
- 15. Jehiel, P., and Moldovanu, B., 1996, "Strategic Non-Participation", *RAND Journal of Economics*, 27, 1, 84-98.
- 16. Joint Report on the Government Securities Market, January 1992, Department of Treasury, SEC and Board of Governors of the Federal Reserve System.

- Jordan, B. D., and Jordan, S. D., 1996, "Salomon Brothers and the May 1991 Treasury Auction: Analysis of a Market Corner", *Journal of Banking and Finance* 20, 1, 25-40.
- Klemperer, P., 1999, "Auction Theory: A Guide to the Literature," Journal of Economic Surveys 13, 227-286.
- Kumar, P., and D. J. Seppi, 1992, "Manipulation with "Cash Settlement", Journal of Finance 47, 1485-1502.
- 20. Kyle, A., 1984, "A Theory of Futures Market Manipulations", in R. Anderson, ed.: *The Industrial Organization of Futures Markets*, Lexington Books, Lexington, Mass.
- 21. Kyle, A., 1989, "Informed Speculation with Imperfect Competition," *Review of Economic Studies* 56, 317-356.
- 22. Malvey, P.F., and C.M. Archibald, 1998, "Uniform-Price Auctions: Update of the Treasury Experience," Office of Market Finance, U.S. Treasury, Washington, D.C.
- Milgrom, P.R., and R. J. Weber, 1982, "A Theory of Auctions and Competitive Bidding", *Econometrica* 50, 1089-1122.
- 24. Nyborg, K.G., 2001, "Underpricing and Market Power in Uniform Price Auctions", working paper, London Business School.
- Nyborg, K.G., K. Rydqvist, and S. Sundaresan, 2001, "Bidder Behavior in Multi-Unit Auctions: Evidence from Swedish Treasury Auctions", *Journal of Political Economy*, forthcoming.
- Nyborg, K.G., and I. A. Strebulaev, 2001, "Collateral and Short Squeezing of Liquidity in Fixed Rate Tenders", *Journal of International Money and Finance* 20, 769-792.
- Nyborg, K.G., and S. Sundaresan, 1996, "Discriminatory versus Uniform Treasury Auctions: Evidence from When-Issued Transactions", *Journal of Financial Economics* 42, 63-104.
- 28. Olsen Jr, M., and R. Zeckhauser, 1966, "An Economic Theory of Alliances," *Review of Economics and Statistics* 48, 266-279.
- 29. Salant, S.W., 1984, "A Theory of Futures Market Manipulations: Comment", in R. Anderson, ed.: *The Industrial Organization of Futures Markets*, Lexington Books, Lexington, Mass.
- 30. Shleifer, A., and R. W. Vishny, 1986, "Large Shareholders and Corporate Control", Journal of Political Economy, 94, 461-88.
- 31. Simon, D.P., 1994, "Markups, Quantity Risk, and Bidding Strategies at Treasury Coupon Auctions," *Journal of Financial Economics* 35, 43-62.

- 32. Sundaresan, S., 1994, "An Empirical Analysis of U.S. Treasury Auctions: Implications for Auction and Term Structure Theories", *Journal of Fixed Income* 35-50.
- 33. Sundaresan S., 1997, *Fixed Income Markets and Their Derivatives*, South-Western College Publishing, Cincinnati, Ohio.
- Vickrey, W., 1961, "Counterspeculation, Auctions, and Competitive Sealed Tenders", Journal of Finance 16, 8-37.
- 35. Viswanathan, S. and J.J.D. Wang, 2000, "Auctions with When-Issued Trading: A Model of the U.S. Treasury Markets," working paper, Fuqua School of Business, Duke University.
- Wilson, R., 1979, "Auctions of Shares", Quarterly Journal of Economics 93, 675-689.









Figure 4: Pr{Short Squeeze} as function of  $|y_{1,0}|$  for N>2



#### NOTE DI LAVORO DELLA FONDAZIONE ENI ENRICO MATTEI Fondazione Eni Enrico Mattei Working Paper Series Our working papers are available on the Internet at the following address

Our working papers are available on the Internet at the following addresses:

http://www.feem.it/web/activ/\_wp.html

http://papers.ssrn.com

SUST	1.2002	K. TANO, M.D. FAMINOW, M. KAMUANGA and B. SWALLOW: Using Conjoint Analysis to Estimate Farmers'
		Preferences for Cattle Traits in West Africa
ETA	2.2002	Efrem CASTELNUOVO and Paolo SURICO: What Does Monetary Policy Reveal about Central Bank's
		Preferences?
WAT	3 2002	Duncan KNOWLER and Edward BARBIER: The Economics of a "Mixed Blessing" Effect: A Case Study of the
,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	3.2002	Black Sea
CLIM	4 2002	
CLINI	4.2002	Andreas LOSCHEL: <u>lechnological Change in Economic Models of Environmental Policy: A Survey</u>
VOL	5.2002	Carlo CARRARO and Carmen MARCHIORI: <u>Stable Coalitions</u>
CLIM	6.2002	Marzio GALEOTTI, Alessandro LANZA and Matteo MANERA: Rockets and Feathers Revisited: An International
		Comparison on European Gasoline Markets
ETA	7.2002	Effrosyni DIAMANTOUDI and Effichios S. SARTZETAKIS: Stable International Environmental Agreements: An
		Analytical Approach
KNOW	8.2002	Alain DESDOIGTS: Neoclassical Convergence Versus Technological Catch-up: A Contribution for Reaching a
		Consensus
NRM	9.2002	Giuseppe DI VITA: Renewable Resources and Waste Recycling
KNOW	10 2002	Giorgio BRUNELLO: Is Training More Frequent when Wage Compression is Higher? Evidence from 11
11100	10.2002	European Countries
FTΔ	11 2002	Morelecai KURZ, Hebui IIN and Maurizio MOTOLESE: Endogenous Eluctuations and the Role of Monetary
LIA	11.2002	Dolari
KNOW	12 2002	roncy
KNOW	12.2002	Rever GERLAGH and Marjan W. HOFRES: Escaping Lock-in: The Scope for a Transition towards Sustainable
1014	12 2002	Growth?
NKM	13.2002	Michele MORE 110 and Paolo ROSATO: <u>The Use of Common Property Resources: A Dynamic Model</u>
CLIM	14.2002	Philippe QUIRION: Macroeconomic Effects of an Energy Saving Policy in the Public Sector
CLIM	15.2002	Roberto ROSON: Dynamic and Distributional Effects of Environmental Revenue Recycling Schemes:
		Simulations with a General Equilibrium Model of the Italian Economy
CLIM	16.2002	Francesco RICCI (1): Environmental Policy Growth when Inputs are Differentiated in Pollution Intensity
ETA	17.2002	Alberto PETRUCCI: Devaluation (Levels versus Rates) and Balance of Payments in a Cash-in-Advance
		Economy
Coalition	18.2002	Láz lá KÁCZV (liu). The Care in the Dresence of Externalities
Theory		Laszio A. KOCZI (IIV). <u>The Core in the Presence of Externatities</u>
Network		
Coalition	19 2002	
Theory	17.2002	Steven J. BRAMS, Michael A. JONES and D. Marc KILGOUR (IIV): Single-Peakedness and Disconnected
Network		Coalitions
Coalition	20 2002	
Theorem	20.2002	Guillaume HAERINGER (liv): On the Stability of Cooperation Structures
Theory		
Network	21 2002	
NKM	21.2002	Fausto CAVALLARO and Luigi CIRAOLO: Economic and Environmental Sustainability: A Dynamic Approach
~ ~ ~		<u>in Insular Systems</u>
CLIM	22.2002	Barbara BUCHNER, Carlo CARRARO, Igor CERSOSIMO and Carmen MARCHIORI: <u>Back to Kyoto? US</u>
		Participation and the Linkage between R&D and Climate Cooperation
CLIM	23.2002	Andreas LÖSCHEL and ZhongXIANG ZHANG: The Economic and Environmental Implications of the US
		Repudiation of the Kyoto Protocol and the Subsequent Deals in Bonn and Marrakech
ETA	24.2002	Marzio GALEOTTI, Louis J. MACCINI and Fabio SCHIANTARELLI: Inventories, Employment and Hours
CLIM	25.2002	Hannes EGLI: Are Cross-Country Studies of the Environmental Kuznets Curve Misleading? New Evidence from
		Time Series Data for Germany
ETA	26.2002	Adam B. JAFFE. Richard G. NEWELL and Robert N. STAVINS: Environmental Policy and Technological
		Change
SUST	27 2002	Josenh C COOPER and Giovanni SIGNORELLO: Farmer Premiums for the Voluntary Adoption of
2001	27.2002	Conservation Plans
SUST	28 2002	The ANSEA Network: Towards An Analytical Strategic Environmental Assessment
KNOW	20.2002	Proto SUBICO: Geographic Concentration and Increasing Deturns: a Survey of Evidence
	20 2002	Pohew N. STAUNS: Lossons from the American Experiment with Market Decad Environmental Delicies
LIA	30.2002	NOUER IN, STAVING, LESSONS NOM THE AMERICAN EXPERIMENT WITH MARKET-DASEU ENVIRONMENTAL POLICIES

NRM	31.2002	Carlo GIUPPONI and Paolo ROSATO: Multi-Criteria Analysis and Decision-Support for Water Management at
		the Catchment Scale: An Application to Diffuse Pollution Control in the Venice Lagoon
NRM	32.2002	Robert N. STAVINS: National Environmental Policy During the Clinton Years
KNOW	33.2002	A. SOUBEYRAN and H. STAHN : Do Investments in Specialized Knowledge Lead to Composite Good
		Industries?
KNOW	34.2002	G. BRUNELLO, M.L. PARISI and Daniela SONEDDA: Labor Taxes, Wage Setting and the Relative Wage
		Effect
CLIM	35.2002	C. BOEMARE and P. QUIRION (lv): Implementing Greenhouse Gas Trading in Europe: Lessons from
		Economic Theory and International Experiences
CLIM	36.2002	T.TIETENBERG (Iv): The Tradable Permits Approach to Protecting the Commons: What Have We Learned?
CLIM	37.2002	K. REHDANZ and R.J.S. TOL (IV): On National and International Trade in Greenhouse Gas Emission Permits
CLIM	38.2002	C. FISCHER (lv): Multinational Taxation and International Emissions Trading
SUST	39.2002	G. SIGNORELLO and G. PAPPALARDO: Farm Animal Biodiversity Conservation Activities in Europe under
		the Framework of Agenda 2000
NRM	40.2002	S. M. CAVANAGH, W. M. HANEMANN and R. N. STAVINS: Muffled Price Signals: Household Water Demand
		under Increasing-Block Prices
NRM	41.2002	A. J. PLANTINGA. R. N. LUBOWSKI and R. N. STAVINS: The Effects of Potential Land Development on
		Agricultural Land Prices
CLIM	42 2002	C OHI (1vi): Inducing Environmental Co-operation by the Design of Emission Permits
CLIM	43 2002	L EVCKMANS, D. VAN REGEMONTER and V. VAN STEENRERGHE (191). Is Kyoto Fatally Flawed? An
CLIW	45.2002	Analysis with MacGEM
CLIM	44 2002	Analysis with MacOEM A ANTOCI and S DORCHESI (hei): Working Too Much in a Dolluted World: A North South Evolutionomy
CLIM	44.2002	A. ANTOCI and S. BORGHESI (W). Working Too Much in a Ponuted world. A North-South Evolutionary
	45 2002	Model D. C. EREDRIKSSON I. I. A. UST ID: INHURPET(I.). Chainethe Southerteche Statement
EIA	45.2002	P. G. FREDRIKSSON, Johan A. LIST and Daniel MILLIMET (IVI): Chasing the Smokestack: Strategic
		Policymaking with Multiple Instruments
ETA	46.2002	Z. YU (Ivi): <u>A Theory of Strategic Vertical DFI and the Missing Pollution-Haven Effect</u>
SUST	47.2002	Y. H. FARZIN: Can an Exhaustible Resource Economy Be Sustainable?
SUST	48.2002	Y. H. FARZIN: Sustainability and Hamiltonian Value
KNOW	49.2002	C. PIGA and M. VIVARELLI: Cooperation in R&D and Sample Selection
Coalition	50.2002	M. SERTEL and A. SLINKO (liv): Ranking Committees, Words or Multisets
Theory		
Network		
Coalition	51.2002	Sergio CURRARINI (liv): Stable Organizations with Externalities
Theory		
Theory Network		
Theory Network FTA	52 2002	Robert N_STAVINS: Experience with Market-Based Policy Instruments
Theory Network ETA	52.2002 53.2002	Robert N. STAVINS: Experience with Market-Based Policy Instruments
Theory Network ETA ETA	52.2002 53.2002	Robert N. STAVINS: Experience with Market-Based Policy Instruments C.C. JAEGER, M. LEIMBACH, C. CARRARO, K. HASSELMANN, J.C. HOURCADE, A. KEELER and R. KLEIN (Jiii): Integrated Assessment Modeling: Modules for Cooperation
Theory Network ETA ETA	52.2002 53.2002	Robert N. STAVINS: Experience with Market-Based Policy Instruments C.C. JAEGER, M. LEIMBACH, C. CARRARO, K. HASSELMANN, J.C. HOURCADE, A. KEELER and R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation Scott BAPPETT (liji): Towards a Patter Climate Treaty
Theory Network ETA ETA CLIM	52.2002 53.2002 54.2002	Robert N. STAVINS: Experience with Market-Based Policy Instruments C.C. JAEGER, M. LEIMBACH, C. CARRARO, K. HASSELMANN, J.C. HOURCADE, A. KEELER and R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation Scott BARRETT (liii): Towards a Better Climate Treaty Biokard C. NEWLL, and Bokert M. STAVING: Cost Haterogeneity and the Potential Society Form Market
Theory Network ETA ETA CLIM ETA	52.2002 53.2002 54.2002 55.2002	Robert N. STAVINS: Experience with Market-Based Policy Instruments C.C. JAEGER, M. LEIMBACH, C. CARRARO, K. HASSELMANN, J.C. HOURCADE, A. KEELER and R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation Scott BARRETT (liii): Towards a Better Climate Treaty Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Decided Delivier
Theory Network ETA ETA CLIM ETA	52.2002 53.2002 54.2002 55.2002	Robert N. STAVINS: Experience with Market-Based Policy Instruments C.C. JAEGER, M. LEIMBACH, C. CARRARO, K. HASSELMANN, J.C. HOURCADE, A. KEELER and R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation Scott BARRETT (liii): Towards a Better Climate Treaty Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies
Theory Network ETA ETA CLIM ETA SUST	52.2002 53.2002 54.2002 55.2002 56.2002	Robert N. STAVINS: Experience with Market-Based Policy Instruments C.C. JAEGER, M. LEIMBACH, C. CARRARO, K. HASSELMANN, J.C. HOURCADE, A. KEELER and R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation Scott BARRETT (liii): Towards a Better Climate Treaty Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs
Theory Network ETA ETA CLIM ETA SUST SUST	52.2002 53.2002 54.2002 55.2002 56.2002 57.2002	Robert N. STAVINS: Experience with Market-Based Policy Instruments         C.C. JAEGER, M. LEIMBACH, C. CARRARO, K. HASSELMANN, J.C. HOURCADE, A. KEELER and         R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation         Scott BARRETT (liii): Towards a Better Climate Treaty         Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market-         Based Policies         Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs         Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of
Theory Network ETA ETA CLIM ETA SUST SUST	52.2002 53.2002 54.2002 55.2002 56.2002 57.2002	Robert N. STAVINS: Experience with Market-Based Policy Instruments         C.C. JAEGER, M. LEIMBACH, C. CARRARO, K. HASSELMANN, J.C. HOURCADE, A. KEELER and         R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation         Scott BARRETT (liii): Towards a Better Climate Treaty         Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market-         Based Policies         Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs         Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests
Theory Network ETA ETA CLIM ETA SUST SUST SUST	52.2002 53.2002 54.2002 55.2002 56.2002 57.2002 58.2002	Robert N. STAVINS: Experience with Market-Based Policy Instruments         C.C. JAEGER, M. LEIMBACH, C. CARRARO, K. HASSELMANN, J.C. HOURCADE, A. KEELER and         R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation         Scott BARRETT (liii): Towards a Better Climate Treaty         Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market-         Based Policies         Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs         Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of         Success and Failures in Implementation of Perspective Quests         Vladimir KOTOV (lvii): Policy in Transition: New Framework for Russia's Climate Policy
Theory Network ETA ETA CLIM ETA SUST SUST SUST SUST	52.2002 53.2002 54.2002 55.2002 56.2002 57.2002 58.2002 59.2002	Robert N. STAVINS: Experience with Market-Based Policy Instruments         C.C. JAEGER, M. LEIMBACH, C. CARRARO, K. HASSELMANN, J.C. HOURCADE, A. KEELER and         R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation         Scott BARRETT (liii): Towards a Better Climate Treaty         Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market-         Based Policies         Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs         Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of         Success and Failures in Implementation of Perspective Quests         Vladimir KOTOV (lvii): Policy in Transition: New Framework for Russia's Climate Policy         Fanny MISSFELDT and Arturo VILLAVICENCO (lvii): How Can Economies in Transition Pursue Emissions
Theory Network ETA ETA CLIM ETA SUST SUST SUST SUST	52.2002 53.2002 54.2002 55.2002 56.2002 57.2002 58.2002 59.2002	Robert N. STAVINS: Experience with Market-Based Policy Instruments         C.C. JAEGER, M. LEIMBACH, C. CARRARO, K. HASSELMANN, J.C. HOURCADE, A. KEELER and         R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation         Scott BARRETT (liii): Towards a Better Climate Treaty         Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market-         Based Policies         Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs         Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of         Success and Failures in Implementation of Perspective Quests         Vladimir KOTOV (lvii): Policy in Transition: New Framework for Russia's Climate Policy         Fanny MISSFELDT and Arturo VILLAVICENCO (lvii): How Can Economies in Transition Pursue Emissions         Trading or Joint Implementation?
Theory Network ETA ETA CLIM ETA SUST SUST SUST SUST VOL	52.2002 53.2002 54.2002 55.2002 56.2002 57.2002 58.2002 59.2002 60.2002	<ul> <li>Robert N. STAVINS: Experience with Market-Based Policy Instruments</li> <li>C.C. JAEGER, M. LEIMBACH, C. CARRARO, K. HASSELMANN, J.C. HOURCADE, A. KEELER and</li> <li>R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation</li> <li>Scott BARRETT (liii): Towards a Better Climate Treaty</li> <li>Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market-</li> <li>Based Policies</li> <li>Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs</li> <li>Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of</li> <li>Success and Failures in Implementation of Perspective Quests</li> <li>Vladimir KOTOV (lvii): Policy in Transition: New Framework for Russia's Climate Policy</li> <li>Fanny MISSFELDT and Arturo VILLAVICENCO (lvii): How Can Economies in Transition Pursue Emissions</li> <li>Trading or Joint Implementation?</li> <li>Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staying Together</li> </ul>
Theory Network ETA ETA CLIM ETA SUST SUST SUST SUST VOL	52.2002 53.2002 54.2002 55.2002 56.2002 57.2002 58.2002 59.2002 60.2002	<ul> <li>Robert N. STAVINS: Experience with Market-Based Policy Instruments</li> <li>C.C. JAEGER, M. LEIMBACH, C. CARRARO, K. HASSELMANN, J.C. HOURCADE, A. KEELER and</li> <li>R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation</li> <li>Scott BARRETT (liii): Towards a Better Climate Treaty</li> <li>Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market-</li> <li>Based Policies</li> <li>Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs</li> <li>Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of</li> <li>Success and Failures in Implementation of Perspective Quests</li> <li>Vladimir KOTOV (lvii): Policy in Transition: New Framework for Russia's Climate Policy</li> <li>Fanny MISSFELDT and Arturo VILLAVICENCO (lvii): How Can Economies in Transition Pursue Emissions</li> <li>Trading or Joint Implementation?</li> <li>Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staving Together</li> <li>or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary</li> </ul>
Theory Network ETA ETA CLIM ETA SUST SUST SUST SUST VOL	52.2002 53.2002 54.2002 55.2002 56.2002 57.2002 58.2002 59.2002 60.2002	<ul> <li>Robert N. STAVINS: Experience with Market-Based Policy Instruments</li> <li>C.C. JAEGER, M. LEIMBACH, C. CARRARO, K. HASSELMANN, J.C. HOURCADE, A. KEELER and</li> <li>R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation</li> <li>Scott BARRETT (liii): Towards a Better Climate Treaty</li> <li>Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market-</li> <li>Based Policies</li> <li>Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs</li> <li>Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of</li> <li>Success and Failures in Implementation of Perspective Quests</li> <li>Vladimir KOTOV (lvii): Policy in Transition: New Framework for Russia's Climate Policy</li> <li>Fanny MISSFELDT and Arturo VILLAVICENCO (lvii): How Can Economies in Transition Pursue Emissions</li> <li>Trading or Joint Implementation?</li> <li>Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staving Together</li> <li>or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary</li> </ul>
Theory Network ETA ETA CLIM ETA SUST SUST SUST SUST VOL ETA	52.2002 53.2002 54.2002 55.2002 56.2002 57.2002 58.2002 59.2002 60.2002 61.2002	Robert N. STAVINS: Experience with Market-Based Policy Instruments         C.C. JAEGER, M. LEIMBACH, C. CARRARO, K. HASSELMANN, J.C. HOURCADE, A. KEELER and         R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation         Scott BARRETT (liii): Towards a Better Climate Treaty         Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market-         Based Policies         Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs         Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of         Success and Failures in Implementation of Perspective Quests         Vladimir KOTOV (lvii): Policy in Transition: New Framework for Russia's Climate Policy         Fanny MISSFELDT and Arturo VILLAVICENCO (lvii): How Can Economies in Transition Pursue Emissions         Trading or Joint Implementation?         Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staying Together         or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary         Union         Robert N. STAVINS, Alexander F.WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic
Theory Network ETA ETA CLIM ETA SUST SUST SUST SUST VOL ETA	52.2002 53.2002 54.2002 55.2002 56.2002 57.2002 58.2002 59.2002 60.2002 61.2002	Robert N. STAVINS: Experience with Market-Based Policy Instruments         C.C. JAEGER, M. LEIMBACH, C. CARRARO, K. HASSELMANN, J.C. HOURCADE, A. KEELER and         R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation         Scott BARRETT (liii): Towards a Better Climate Treaty         Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market-         Based Policies         Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs         Vladimir KOTOV and Elena NIKITINA (Ivii): Reorganisation of Environmental Policy in Russia: The Decade of         Success and Failures in Implementation of Perspective Quests         Vladimir KOTOV (Ivii): Policy in Transition: New Framework for Russia's Climate Policy         Fanny MISSFELDT and Arturo VILLAVICENCO (Ivii): How Can Economies in Transition Pursue Emissions         Trading or Joint Implementation?         Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staving Together         or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary         Union         Robert N. STAVINS, Alexander F.WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic         Terms: Dynamic Efficiency Plus Intergenerational Equity
Theory Network ETA ETA CLIM ETA SUST SUST SUST SUST VOL ETA PRIV	52.2002 53.2002 54.2002 55.2002 56.2002 57.2002 58.2002 59.2002 60.2002 61.2002 62.2002	Robert N. STAVINS: Experience with Market-Based Policy Instruments         C.C. JAEGER, M. LEIMBACH, C. CARRARO, K. HASSELMANN, J.C. HOURCADE, A. KEELER and         R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation         Scott BARRETT (liii): Towards a Better Climate Treaty         Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market-         Based Policies         Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs         Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of         Success and Failures in Implementation of Perspective Quests         Vladimir KOTOV (lvii): Policy in Transition: New Framework for Russia's Climate Policy         Fanny MISSFELDT and Arturo VILLAVICENCO (lvii): How Can Economics in Transition Pursue Emissions         Trading or Joint Implementation?         Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staying Together         or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary         Union         Robert N. STAVINS, Alexander F. WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic         Terms: Dynamic Efficiency Plus Intergenerational Equity         Carlo CAPUANO: Demand Growth, Entry and Collusion Sustainability
Theory Network ETA ETA CLIM ETA SUST SUST SUST SUST VOL ETA PRIV PRIV	52.2002 53.2002 54.2002 55.2002 56.2002 57.2002 58.2002 59.2002 60.2002 61.2002 62.2002 63.2002	Robert N. STAVINS: Experience with Market-Based Policy Instruments         C.C. JAEGER, M. LEIMBACH, C. CARRARO, K. HASSELMANN, J.C. HOURCADE, A. KEELER and         R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation         Scott BARRETT (liii): Towards a Better Climate Treaty         Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market-         Based Policies         Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs         Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of         Success and Failures in Implementation of Perspective Quests         Vladimir KOTOV (lvii): Policy in Transition: New Framework for Russia's Climate Policy         Fanny MISSFELDT and Arturo VILLAVICENCO (lvii): How Can Economies in Transition Pursue Emissions         Trading or Joint Implementation?         Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staving Together         or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary         Union         Robert N. STAVINS, Alexander F.WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic         Terms: Dynamic Efficiency Plus Intergenerational Equity         Carlo CAPUANO: Demand Growth, Entry and Collusion Sustainability         Federico MUNARI and Raffaele ORIANI: Privatization and R&D Performance: An Empirical Analysis Based on
Theory Network ETA ETA CLIM ETA SUST SUST SUST SUST VOL ETA PRIV PRIV	52.2002 53.2002 54.2002 55.2002 56.2002 57.2002 58.2002 59.2002 60.2002 61.2002 62.2002 63.2002	Robert N. STAVINS: Experience with Market-Based Policy Instruments         C.C. JAEGER, M. LEIMBACH, C. CARRARO, K. HASSELMANN, J.C. HOURCADE, A. KEELER and         R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation         Scott BARRETT (liii): Towards a Better Climate Treaty         Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies         Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs         Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of         Success and Failures in Implementation of Perspective Quests         Vladimir KOTOV (lvii): Policy in Transition: New Framework for Russia's Climate Policy         Fanny MISSFELDT and Arturo VILLAVICENCO (lvii): How Can Economies in Transition Pursue Emissions         Trading or Joint Implementation?         Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staving Together         or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary         Union         Robert N. STAVINS, Alexander F.WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic         Terms: Dynamic Efficiency Plus Intergenerational Equity         Carlo CAPUANO: Demand Growth, Entry and Collusion Sustainability         Federico MUNARI and Raffaele ORIANI: Privatization and R&D Performance: An Empirical Analysis Based on
Theory Network ETA ETA CLIM ETA SUST SUST SUST VOL ETA PRIV PRIV	52.2002 53.2002 54.2002 55.2002 56.2002 57.2002 58.2002 59.2002 60.2002 61.2002 61.2002 63.2002 63.2002	Robert N. STAVINS: Experience with Market-Based Policy Instruments         C.C. JAEGER, M. LEIMBACH, C. CARRARO, K. HASSELMANN, J.C. HOURCADE, A. KEELER and         R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation         Scott BARRETT (liii): Towards a Better Climate Treaty         Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market-         Based Policies         Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs         Vladimir KOTOV and Elena NIKITINA (Ivii): Reorganisation of Environmental Policy in Russia: The Decade of         Success and Failures in Implementation of Perspective Quests         Vladimir KOTOV (Ivii): Policy in Transition: New Framework for Russia's Climate Policy         Fanny MISSFELDT and Arturo VILLAVICENCO (Ivii): How Can Economies in Transition Pursue Emissions         Trading or Joint Implementation?         Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staying Together         or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary         Union         Robert N. STAVINS, Alexander F.WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic         Terms: Dynamic Efficiency Plus Intergenerational Equity         Carlo CAPUANO: Demand Growth, Entry and Collusion Sustainability         Federico MUNARI and Raffaele ORIANI: Privatization and R&D Performance: An Empirical Analysis Based on <t< td=""></t<>
Theory Network ETA ETA CLIM ETA SUST SUST SUST VOL ETA PRIV PRIV	52.2002 53.2002 54.2002 55.2002 56.2002 57.2002 58.2002 59.2002 60.2002 61.2002 61.2002 63.2002 64.2002	Robert N. STAVINS: Experience with Market-Based Policy Instruments         C.C. JAEGER, M. LEIMBACH, C. CARRARO, K. HASSELMANN, J.C. HOURCADE, A. KEELER and         R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation         Scott BARRETT (liii): Towards a Better Climate Treaty         Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market-         Based Policies         Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs         Vladimir KOTOV and Elena NIKITINA (Ivii): Reorganisation of Environmental Policy in Russia; The Decade of         Success and Failures in Implementation of Perspective Quests         Vladimir KOTOV (Ivii): Policy in Transition: New Framework for Russia's Climate Policy         Fanny MISSFELDT and Arturo VILLAVICENCO (Ivii): How Can Economics in Transition Pursue Emissions         Trading or Joint Implementation?         Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staying Together         or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary         Union         Robert N. STAVINS, Alexander F. WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic         Terms: Dynamic Efficiency Plus Intergenerational Equity         Carlo CAPUANO: Demand Growth, Entry and Collusion Sustainability         Federico MUNARI and Raffaele ORIANI: Privatization and R&D Performance: An Empirical Analysis Based on Tobin's Q
Theory Network ETA ETA CLIM ETA SUST SUST SUST VOL ETA PRIV PRIV PRIV	52.2002 53.2002 54.2002 55.2002 56.2002 57.2002 58.2002 59.2002 60.2002 61.2002 61.2002 63.2002 64.2002 64.2002	Robert N. STAVINS: Experience with Market-Based Policy Instruments         C.C. JAEGER, M. LEIMBACH, C. CARRARO, K. HASSELMANN, J.C. HOURCADE, A. KEELER and         R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation         Scott BARRETT (liii): Towards a Better Climate Treaty         Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies         Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs         Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of         Success and Failures in Implementation of Perspective Quests         Vladimir KOTOV (lvii): Policy in Transition: New Framework for Russia's Climate Policy         Fanny MISSFELDT and Arturo VILLAVICENCO (lvii): How Can Economies in Transition Pursue Emissions         Trading or Joint Implementation?         Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staying Together         or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary         Union         Robert N. STAVINS, Alexander F. WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic         Terms: Dynamic Efficiency Plus Intergenerational Equity         Carlo CAPUANO: Demand Growth, Entry and Collusion Sustainability         Federico MUNARI and Raffaele ORIANI: Privatization and R&D Performance: An Empirical Analysis Based on Tobin's Q         Federi
Theory Network ETA ETA CLIM ETA SUST SUST SUST VOL ETA PRIV PRIV PRIV SUST	52.2002 53.2002 54.2002 55.2002 56.2002 57.2002 58.2002 59.2002 60.2002 61.2002 63.2002 64.2002 65.2002	Robert N. STAVINS: Experience with Market-Based Policy Instruments         C.C. JAEGER, M. LEIMBACH, C. CARRARO, K. HASSELMANN, J.C. HOURCADE, A. KEELER and         R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation         Scott BARRETT (liii): Towards a Better Climate Treaty         Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market-         Based Policies         Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs         Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of         Success and Failures in Implementation of Perspective Quests         Vladimir KOTOV (lvii): Policy in Transition: New Framework for Russia's Climate Policy         Fanny MISSFELDT and Arturo VILLAVICENCO (lvii): How Can Economies in Transition Pursue Emissions         Trading or Joint Implementation?         Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staying Together         or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary         Union         Robert N. STAVINS, Alexander F. WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic         Terms: Dynamic Efficiency Plus Intergenerational Equity         Carlo CAPUANO: Demand Growth, Entry and Collusion Sustainability         Federico MUNARI and Raffaele ORIANI: Privatization and R&D Performance: An Empirical Analysis Based on      <
Theory Network ETA ETA CLIM ETA SUST SUST SUST VOL ETA PRIV PRIV PRIV SUST	52.2002 53.2002 54.2002 55.2002 55.2002 57.2002 58.2002 59.2002 60.2002 61.2002 61.2002 63.2002 64.2002 65.2002	Robert N. STAVINS: Experience with Market-Based Policy Instruments         C.C. JAEGER, M. LEIMBACH, C. CARRARO, K. HASSELMANN, J.C. HOURCADE, A. KEELER and         R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation         Scott BARRETT (liii): Towards a Better Climate Treaty         Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market-         Based Policies         Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs         Vladimir KOTOV and Elena NIKITINA (Ivii): Reorganisation of Environmental Policy in Russia: The Decade of         Success and Failures in Implementation of Perspective Quests         Vladimir KOTOV (Ivii): Policy in Transition: New Framework for Russia's Climate Policy         Fanny MISSFELDT and Arturo VILLAVICENCO (Ivii): How Can Economics in Transition Pursue Emissions         Trading or Joint Implementation?         Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staving Together         or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary         Union         Robert N. STAVINS, Alexander F.WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic         Terms: Dynamic Efficiency Plus Intergenerational Equity         Carlo CAPUANO: Demand Growth, Entry and Collusion Sustainability         Federico MUNARI and Raffaele ORIANI: Privatization and R&D Performance: An Empirical Analysis Based on <t< td=""></t<>
Theory Network ETA ETA CLIM ETA SUST SUST SUST VOL ETA PRIV PRIV PRIV SUST ETA PDIV	52.2002 53.2002 54.2002 55.2002 55.2002 57.2002 58.2002 59.2002 60.2002 61.2002 61.2002 63.2002 64.2002 64.2002 65.2002 65.2002	Robert N. STAVINS: Experience with Market-Based Policy Instruments         C.C. JAEGER, M. LEIMBACH, C. CARRARO, K. HASSELMANN, J.C. HOURCADE, A. KEELER and         R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation         Scott BARRETT (liii): Towards a Better Climate Treaty         Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market-         Based Policies         Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs         Vladimir KOTOV and Elena NIKITINA (Ivii): Reorganisation of Environmental Policy in Russia: The Decade of         Success and Failures in Implementation of Perspective Quests         Vladimir KOTOV (Ivii): Policy in Transition: New Framework for Russia's Climate Policy         Fanny MISSFELDT and Arturo VILLAVICENCO (Ivii): How Can Economies in Transition Pursue Emissions         Trading or Joint Implementation?         Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staving Together         or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary         Union         Robert N. STAVINS, Alexander F. WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic         Terms: Dynamic Efficiency Plus Intergenerational Equity         Carlo CAPUANO: Demand Growth, Entry and Collusion Sustainability         Federico MUNARI and Raffaele ORIANI: Privatization and R&D Performance: An Empirical Analysis Based on      <
Theory Network ETA ETA CLIM ETA SUST SUST SUST VOL ETA PRIV PRIV SUST ETA PRIV	52.2002 53.2002 54.2002 55.2002 55.2002 57.2002 58.2002 59.2002 60.2002 61.2002 61.2002 63.2002 64.2002 64.2002 65.2002 65.2002	Robert N. STAVINS: Experience with Market-Based Policy Instruments         C.C. JAEGER, M. LEIMBACH, C. CARRARO, K. HASSELMANN, J.C. HOURCADE, A. KEELER and         R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation         Scott BARRETT (liii): Towards a Better Climate Treaty         Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies         Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs         Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of         Success and Failures in Implementation of Perspective Quests         Vladimir KOTOV (lvii): Policy in Transition: New Framework for Russia's Climate Policy         Fanny MISSFELDT and Arturo VILLAVICENCO (lvii): How Can Economics in Transition Pursue Emissions         Trading or Joint Implementation?         Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staying Together         or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary         Union         Robert N. STAVINS, Alexander F. WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic         Terms: Dynamic Efficiency Plus Intergenerational Equity         Carlo CAPUANO: Demand Growth, Entry and Collusion Sustainability         Federico MUNARI and Raffaele ORIANI: Privatization and R&D Performance: An Empirical Analysis Based on         Tobin's Q
Theory Network ETA ETA CLIM ETA SUST SUST SUST VOL ETA PRIV PRIV SUST ETA PRIV	52.2002 53.2002 54.2002 55.2002 55.2002 57.2002 58.2002 59.2002 60.2002 61.2002 61.2002 63.2002 64.2002 64.2002 65.2002 65.2002	Robert N. STAVINS: Experience with Market-Based Policy Instruments         C.C. JAEGER, M. LEIMBACH, C. CARRARO, K. HASSELMANN, J.C. HOURCADE, A. KEELER and         R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation         Scott BARRETT (liii): Towards a Better Climate Treaty         Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market-         Based Policies         Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs         Vladimir KOTOV and Elena NIKITINA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of         Success and Failures in Implementation of Perspective Quests         Vladimir KOTOV (lvii): Policy in Transition: New Framework for Russia's Climate Policy         Fanny MISSFELDT and Arturo VILLAVICENCO (lvii): How Can Economics in Transition Pursue Emissions         Trading or Joint Implementation?         Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staying Together         or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary         Ulnion         Robert N. STAVINS, Alexander F.WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic         Terms: Dynamic Efficiency Plus Intergenerational Equity         Carlo CAPUANO: Demand Growth, Entry and Collusion Sustainability         Federico MUNARI and Raffaele ORIANI: Privatization and R&D Performance: An Empirical Analysis Based on      <
Theory Network ETA ETA CLIM ETA SUST SUST SUST VOL ETA PRIV PRIV PRIV SUST ETA PRIV	<ul> <li>52.2002</li> <li>53.2002</li> <li>54.2002</li> <li>55.2002</li> <li>55.2002</li> <li>56.2002</li> <li>57.2002</li> <li>58.2002</li> <li>59.2002</li> <li>60.2002</li> <li>61.2002</li> <li>62.2002</li> <li>64.2002</li> <li>64.2002</li> <li>65.2002</li> <li>66.2002</li> <li>67.2002</li> <li>68.2002</li> <li>68.2002</li> </ul>	Robert N. STAVINS: Experience with Market-Based Policy Instruments         C.C. JAEGER, M. LEIMBACH, C. CARRARO, K. HASSELMANN, J.C. HOURCADE, A. KEELER and         R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation         Scott BARRETT (liii): Towards a Better Climate Treaty         Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies         Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs         Vladimir KOTOV and Elena NIKITINA (Wi): Reorganisation of Environmental Policy in Russia: The Decade of         Success and Failures in Implementation of Perspective Quests         Vladimir KOTOV (Wi): Policy in Transition: New Framework for Russia's Climate Policy         Fanny MISSFELDT and Arturo VILLAVICENCO (Ivii): How Can Economies in Transition Pursue Emissions         Trading or Joint Implementation?         Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staving Together         Or or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary         Union         Robert N. STAVINS, Alexander F.WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic         Terms: Dynamic Efficiency Plus Intergenerational Equity         Federico MUNARI and Raffaele ORIANI: Privatization and R&D Performance: An Empirical Analysis Based on         Tobin's Q         Federico MUNARI and Maurizio SOBRERO: The Effects of Privatization on R&D Inv
Theory Network ETA ETA CLIM ETA SUST SUST SUST VOL ETA PRIV PRIV PRIV SUST ETA PRIV CLIM CLIM	<ul> <li>52.2002</li> <li>53.2002</li> <li>54.2002</li> <li>55.2002</li> <li>55.2002</li> <li>56.2002</li> <li>57.2002</li> <li>58.2002</li> <li>59.2002</li> <li>60.2002</li> <li>61.2002</li> <li>61.2002</li> <li>62.2002</li> <li>64.2002</li> <li>64.2002</li> <li>65.2002</li> <li>66.2002</li> <li>67.2002</li> <li>68.2002</li> <li>69.2002</li> </ul>	Robert N. STAVINS: Experience with Market-Based Policy Instruments         C.C. JAEGER, M. LEIMBACH, C. CARRARO, K. HASSELMANN, J.C. HOURCADE, A. KEELER and         R. KLEIN (Iiii): Integrated Assessment Modeling: Modules for Cooperation         Scott BARRETT (Iiii): Towards a Better Climate Treaty         Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market-         Based Policies         Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs         Vladimir KOTOV and Elena NIKITINA (Ivii): Reorganisation of Environmental Policy in Russia: The Decade of         Success and Failures in Implementation of Perspective Quests         Vladimir KOTOV (Ivii): Policy in Transition: New Framework for Russia's Climate Policy         Fnany MISSFELDT and Arturo VILLAVICENCO (Ivii): How Can Economies in Transition Pursue Emissions         Trading or Joint Implementation?         Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staying Together         or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary         Ulnion         Robert N. STAVINS, Alexander F. WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic         Terms: Dynamic Efficiency Plus Intergenerational Equity         Carlo CAPUANO: Demand Growth, Entry and Collusion Sustainability         Federico MUNARI and Raffaele ORIANI: Privatization and R&D Performance: An Empirical Analysis Based on
Theory Network ETA ETA CLIM ETA SUST SUST SUST VOL ETA PRIV PRIV PRIV SUST ETA PRIV CLIM CLIM SUST	<ul> <li>52.2002</li> <li>53.2002</li> <li>54.2002</li> <li>55.2002</li> <li>55.2002</li> <li>56.2002</li> <li>57.2002</li> <li>58.2002</li> <li>59.2002</li> <li>60.2002</li> <li>61.2002</li> <li>62.2002</li> <li>64.2002</li> <li>64.2002</li> <li>65.2002</li> <li>66.2002</li> <li>67.2002</li> <li>68.2002</li> <li>69.2002</li> <li>70.2002</li> </ul>	Robert N. STAVINS: Experience with Market-Based Policy Instruments         C.C. JAEGER, M. LEIMBACH, C. CARRARO, K. HASSELMANN, J.C. HOURCADE, A. KEELER and         R. KLEIN (Iiii): Integrated Assessment Modeling: Modules for Cooperation         Scott BARRETT (Iiii): Towards a Better Climate Treaty         Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market- Based Policies         Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs         Vladimir KOTOV and Elena NIKITINA (Ivii): Reorganisation of Environmental Policy in Russia: The Decade of         Success and Failures in Implementation of Perspective Quests         Vladimir KOTOV (Ivii): Policy in Transition: New Framework for Russia's Climate Policy         Famny MISSFELDT and Arturo VILLAVICENCO (Ivii): How Can Economies in Transition Pursue Emissions         Trading or Joint Implementation?         Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staving Together         or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary         Union         Robert N. STAVINS, Alexander F.WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic         Federico MUNARI and Raffaele ORIANI: Privatization and R&D Performance: An Empirical Analysis Based on         Tobin's Q         Federico MUNARI and Maurizio SOBRERO: The Effects of Privatization on R&D Investments and Patent         Productivity <tr< td=""></tr<>
Theory Network ETA ETA CLIM ETA SUST SUST SUST VOL ETA PRIV PRIV SUST ETA PRIV CLIM CLIM SUST	<ul> <li>52.2002</li> <li>53.2002</li> <li>54.2002</li> <li>55.2002</li> <li>55.2002</li> <li>57.2002</li> <li>57.2002</li> <li>59.2002</li> <li>60.2002</li> <li>61.2002</li> <li>62.2002</li> <li>64.2002</li> <li>65.2002</li> <li>66.2002</li> <li>67.2002</li> <li>68.2002</li> <li>69.2002</li> <li>70.2002</li> </ul>	<ul> <li>Robert N. STAVINS: Experience with Market-Based Policy Instruments</li> <li>C.C. JAEGER, M. LEIMBACH, C. CARRARO, K. HASSELMANN, J.C. HOURCADE, A. KEELER and</li> <li>R. KLEIN (liii): Integrated Assessment Modeling: Modules for Cooperation</li> <li>Scott BARRETT (liii): Towards a Better Climate Treaty</li> <li>Richard G. NEWELL and Robert N. STAVINS: Cost Heterogeneity and the Potential Savings from Market-Based Policies</li> <li>Paolo ROSATO and Edi DEFRANCESCO: Individual Travel Cost Method and Flow Fixed Costs</li> <li>Vladimir KOTOV and Elena NIKITIMA (lvii): Reorganisation of Environmental Policy in Russia: The Decade of Success and Failures in Implementation of Perspective Quests</li> <li>Vladimir KOTOV (lvii): Policy in Transition: New Framework for Russia's Climate Policy</li> <li>Fanny MISSFELDT and Arturo VILLAVICENCO (lvii): How Can Economics in Transition Pursue Emissions</li> <li>Trading or Joint Implementation?</li> <li>Giovanni DI BARTOLOMEO, Jacob ENGWERDA, Joseph PLASMANS and Bas VAN AARLE: Staving Together</li> <li>or Breaking Apart: Policy-Makers' Endogenous Coalitions Formation in the European Economic and Monetary Union</li> <li>Robert N. STAVINS, Alexander F. WAGNER and Gernot WAGNER: Interpreting Sustainability in Economic</li> <li>Terms: Dynamic Efficiency Plus Intergenerational Equity</li> <li>Carlo CAPUANO: Demand Growth, Entry and Collusion Sustainability</li> <li>Federico MUNARI and Raffaele ORIANI: Privatization and R&amp;D Performance: An Empirical Analysis Based on</li> <li>Tobin's Q</li> <li>Federico MUNARI and Maurizio SOBRERO: The Effects of Privatization on R&amp;D Investments and Patent</li> <li>Productivity</li> <li>Barbara K. BUCHNER and Roberto ROSON: Conflicting Perspectives in Trade and Environmental Negotiations</li> <li>Philappe QUIRION: Complying with the Kyoto Protocol under Uncertainty: Taxes or Tradable Permits?</li> <li>Anna AlBERINI, Patrita RIGANTI and Alberto LONGO: Can Peo</li></ul>

NRM	72.2002	Philippe BONTEMS and Pascal FAVARD: Input Use and Capacity Constraint under Uncertainty: The Case of
	72 2002	Irrigation
PRIV	/3.2002	Mohammed OMRAN: <u>The Performance of State-Owned Enterprises and Newly Privatized Firms: Empirical</u>
PRIV	74 2002	Evidence from Egypt Mike RURKART Equato PANUNZL and Andrei SHI EIEER: Family Firms
PRIV	75.2002	<i>Emmanuelle AURIOL, Pierre M. PICARD:</i> Privatizations in Developing Countries and the Government Budget
	/0.2002	Constraint
PRIV	76.2002	Nichole M. CASTATER: Privatization as a Means to Societal Transformation: An Empirical Study of
		Privatization in Central and Eastern Europe and the Former Soviet Union
PRIV	77.2002	Christoph LÜLSFESMANN: Benevolent Government, Managerial Incentives, and the Virtues of Privatization
PRIV	78.2002	Kate BISHOP, Igor FILATOTCHEV and Tomasz MICKIEWICZ: Endogenous Ownership Structure: Factors
		Affecting the Post-Privatisation Equity in Largest Hungarian Firms
PRIV	79.2002	Theodora WELCH and Rick MOLZ: <u>How Does Trade Sale Privatization Work?</u>
DDIV	80 2002	Evidence from the Fixed-Line Telecommunications Sector in Developing Economies
PKIV	80.2002	<i>Alberto R. PETROCCT</i> . Government Debi, Agent Helerogeneity and wealth Displacement in a Small Open
CLIM	81 2002	Timothy SWANSON and Robin MASON (lvi): The Impact of International Environmental Agreements: The Case
CLIM	01.2002	of the Montreal Protocol
PRIV	82.2002	George R.G. CLARKE and Lixin Colin XU: Privatization. Competition and Corruption: How Characteristics of
		Bribe Takers and Payers Affect Bribe Payments to Utilities
PRIV	83.2002	Massimo FLORIO and Katiuscia MANZONI: The Abnormal Returns of UK Privatisations: From Underpricing
		to Outperformance
NRM	84.2002	Nelson LOURENÇO, Carlos RUSSO MACHADO, Maria do ROSÁRIO JORGE and Luís RODRIGUES: An
		Integrated Approach to Understand Territory Dynamics. The Coastal Alentejo (Portugal)
CLIM	85.2002	Peter ZAPFEL and Matti VAINIO (lv): Pathways to European Greenhouse Gas Emissions Trading History and
CL D /	06 0000	Misconceptions
CLIM	86.2002	<i>Pierre COURTOIS</i> : Influence Processes in Climate Change Negotiations: Modelling the Rounds
EIA	87.2002	VIIO FRAGNELLI and Maria Erminia MARINA (IVIII): <u>Environmental Pollution Risk and Insurance</u>
	88.2002	Time COESCHI and Timethy M SWANSON (lyiji): Lost Herizons. The noncooperative management of an
LIA	07.2002	evolutionary biological system
ETA	90.2002	Hans KEIDING (lyiji): Environmental Effects of Consumption: An Approach Using DEA and Cost Sharing
ETA	91.2002	Wietze LISE (Iviii): A Game Model of People's Participation in Forest Management in Northern India
CLIM	92.2002	Jens HORBACH: Structural Change and Environmental Kuznets Curves
ETA	93.2002	Martin P. GROSSKOPF: Towards a More Appropriate Method for Determining the Optimal Scale of Production
		Units
VOL	94.2002	Scott BARRETT and Robert STAVINS: Increasing Participation and Compliance in International Climate Change
CL D /	05 0000	Agreements
CLIM	95.2002	Banu BAYRAMOGLU LISE and Wietze LISE: Climate Change, Environmental NGOs and Public Awareness in
CUM	06 2002	International Structure and Reality Matthian CLACHANT: The Delitical Economy of Emission Tax Design in Environmental Delicy
KNOW	90.2002	Kann ARIGA and Giorgio BRUNELLO: Are the More Educated Receiving More Training? Evidence from
KINOW	77.2002	Thailand
ETA	98.2002	Gianfranco FORTE and Matteo MANERA: Forecasting Volatility in European Stock Markets with Non-linear
2111	2002	GARCH Models
ETA	99.2002	Geoffrey HEAL: Bundling Biodiversity
ETA	100.2002	Geoffrey HEAL, Brian WALKER, Simon LEVIN, Kenneth ARROW, Partha DASGUPTA, Gretchen DAILY, Paul
		EHRLICH, Karl-Goran MALER, Nils KAUTSKY, Jane LUBCHENCO, Steve SCHNEIDER and David
		STARRETT: Genetic Diversity and Interdependent Crop Choices in Agriculture
ETA	101.2002	Geoffrey HEAL: Biodiversity and Globalization
VOL	102.2002	Andreas LANGE: Heterogeneous International Agreements – If per capita emission levels matter
EIA	103.2002	Pierre-Andre JOUVET and Walid OUESLATI: <u>Tax Reform and Public Spending Trade-offs in an Endogenous</u>
ETA	104 2002	Arma ROTTASSO and Alessandre SEMPENELLI: Door Ownership Affect Firms' Efficiency? Banal Date
LIA	104.2002	<i>Annu DOTTASSO unu Alessunuro SEMDENELLI. <u>Does Ownersnip Affect Fillis Efficiency: Fallet Data</u> Evidence on Italy</i>
PRIV	105 2002	Bernardo RORTOLOTTI Frank DE JONG Giovanna NICODANO and Ibolva SCHINDELE: Privatization and
1101	100.2002	Stock Market Liquidity
ETA	106.2002	Haruo IMAI and Mayumi HORIE (Iviii): Pre-Negotiation for an International Emission Reduction Game
PRIV	107.2002	Sudeshna GHOSH BANERJEE and Michael C. MUNGER: Move to Markets? An Empirical Analysis of
		Privatisation in Developing Countries
PRIV	108.2002	Guillaume GIRMENS and Michel GUILLARD: Privatization and Investment: Crowding-Out Effect vs Financial
		Diversification
PRIV	109.2002	Alberto CHONG and Florencio LOPEZ-DE-SILANES: Privatization and Labor Force Restructuring Around the
	110 2002	World New divis CUDT (), Dertial Drivertiantian and Firm Derformer (1)
	110.2002	IVARIATION OUT LAT <u>PARUAL PRIVAUZATION and FIRM PERFORMANCE</u> Examples DEGEORGE Dirk IENTER Alberto MOEL and Pater TUEANO: Solling Company Shares to
1 KIV	111.2002	Reluctant Employees: France Telecom's Experience
		Refue with Employees, Figuree Felecom 5 Experience

PRIV	112.2002	Isaac OTCHERE: Intra-Industry Effects of Privatization Announcements: Evidence from Developed and
		Developing Countries
PRIV	113.2002	Yannis KATSOULAKOS and Elissavet LIKOYANNI: Fiscal and Other Macroeconomic Effects of Privatization
PRIV	114.2002	Guillaume GIRMENS: Privatization, International Asset Trade and Financial Markets
PRIV	115.2002	D. Teia FLOTHO: A Note on Consumption Correlations and European Financial Integration
PRIV	116.2002	Ibolya SCHINDELE and Enrico C. PEROTTI: Pricing Initial Public Offerings in Premature Capital Markets:
		The Case of Hungary
PRIV	1 2003	Gabrialia CHIESA and Giovanna NICODANO. Privatization and Financial Market Development: Theoretical
	1.2005	
DDIV	2 2003	<u>Isolus</u> Ibolus SCHINDELE: Theory of Drivatization in Eastern Europe: Literature Raview
	2.2003	<i>Wates USE Cludis VETMEDET and Disland S. I. TOL:</i> Stratagic Action in the Liberalized Cormon Electricity.
PKIV	3.2003	Weize LISE, Claudia KEMFERT and Richard S.J. TOL: Strategic Action in the Liberarised German Electricity
arn r		Market
CLIM 4.2003 Laura MARSILIANI and Thomas I. RENSTROM: Environmental Policy and Capital Movemen		Laura MARSILIANI and Thomas I. RENSTROM: Environmental Policy and Capital Movements: The Role of
		<u>Government Commitment</u>
KNOW	5.2003	Reyer GERLAGH: Induced Technological Change under Technological Competition
ETA	6.2003	<i>Efrem CASTELNUOVO</i> : <u>Squeezing the Interest Rate Smoothing Weight with a Hybrid Expectations Model</u>
SIEV	7.2003	Anna ALBERINI, Alberto LONGO, Stefania TONIN, Francesco TROMBETTA and Margherita TURVANI: The
		Role of Liability, Regulation and Economic Incentives in Brownfield Remediation and Redevelopment:
		Evidence from Surveys of Developers
NRM	8.2003	Elissaios PAPYRAKIS and Rever GERLAGH: Natural Resources: A Blessing or a Curse?
CLIM	9 2003	A CAPARRÓS J-C PEREAU and T TAZDAÏT. North-South Climate Change Negotiations: a Sequential Game
CLIII	2.2000	with Asymmetric Information
KNOW	10 2003	<u>main response to the maximum</u>
CLIM	11 2003	From CASTELNIOV and Marrie GALEATTIC Learning By Doing vs Learning By Recearching in a Model of
CLIW	11.2003	Elimit CASTELIVOVO un Marzio GALEOTTI. Leaning By Doing vs Leaning By Researching in a Wooder of
KNOW	12 2002	Commander Change Foncy Antarysis
KNOW	12.2003	Carole MAGNAN, Glanmarco OTTAVIANO and Dino PINELLI (eas.): Economic Growth, Innovation, Cultural
www.	12 2002	Diversity: What are we all talking about? A critical survey of the state-or-the-art
KNOW	13.2003	Carole MAIGNAN, Gianmarco OTTAVIANO, Dino PINELLI and Francesco RULLANI (IVIX): BIO-Ecological
		Diversity vs. Socio-Economic Diversity. A Comparison of Existing Measures
KNOW	14.2003	Maddy JANSSENS and Chris STEYAERT (Ivix): Theories of Diversity within Organisation Studies: Debates and
		<u>Future Trajectories</u>
KNOW	15.2003	Tuzin BAYCAN LEVENT, Enno MASUREL and Peter NIJKAMP (lvix): Diversity in Entrepreneurship: Ethnic
		and Female Roles in Urban Economic Life
KNOW	16.2003	Alexandra BITUSIKOVA (lvix): Post-Communist City on its Way from Grey to Colourful: The Case Study from
		Slovakia
KNOW	17.2003	Billy E. VAUGHN and Katarina MLEKOV (lyix): A Stage Model of Developing an Inclusive Community
KNOW	18 2003	Selma van LONDEN and Arie de RULITER (lvix): Managing Diversity in a Glocalizing World
Coalition	19 2003	Service CURRARINI: On the Stability of Hierarchies in Games with Externalities
Theory	17.2005	Sergio Contentini, on the Submy of Includences in Games with Externations
Network		
DDIV	20 2003	Giasome CALTOLAPL and Alessandro PAVAN (lux): Monopoly with Possile
	20.2003	Glacomo CALZOLANI and Alessandro FAVAIV (IVX). Monopory with Kesale
PKIV	21.2003	Characteristic Revenue of the second se
		Principle, Efficiency, Full Surplus Extraction and Information Acquisition
PRIV	22.2003	Marco LiCalzi and Alessandro PAVAN (Ivx): <u>Tilting the Supply Schedule to Enhance Competition in Uniform-</u>
		Price Auctions
PRIV	23.2003	David ETTINGER (lvx): Bidding among Friends and Enemies
PRIV	24.2003	Hannu VARTIAINEN (lvx): Auction Design without Commitment
PRIV	25.2003	Matti KELOHARJU, Kjell G. NYBORG and Kristian RYDQVIST (lvx): Strategic Behavior and Underpricing in
		Uniform Price Auctions: Evidence from Finnish Treasury Auctions
PRIV	26.2003	Christine A. PARLOUR and Uday RAJAN (lvx): <u>Rationing in IPOs</u>
PRIV	27.2003	Kjell G. NYBORG and Ilya A. STREBULAEV (Ivx): Multiple Unit Auctions and Short Squeezes

(1) This paper was presented at the Workshop "Growth, Environmental Policies and Sustainability" organised by the Fondazione Eni Enrico Mattei, Venice, June 1, 2001

(li) This paper was presented at the Fourth Toulouse Conference on Environment and Resource Economics on "Property Rights, Institutions and Management of Environmental and Natural Resources", organised by Fondazione Eni Enrico Mattei, IDEI and INRA and sponsored by MATE, Toulouse, May 3-4, 2001

(lii) This paper was presented at the International Conference on "Economic Valuation of Environmental Goods", organised by Fondazione Eni Enrico Mattei in cooperation with CORILA, Venice, May 11, 2001

(liii) This paper was circulated at the International Conference on "Climate Policy – Do We Need a New Approach?", jointly organised by Fondazione Eni Enrico Mattei, Stanford University and Venice International University, Isola di San Servolo, Venice, September 6-8, 2001

(liv) This paper was presented at the Seventh Meeting of the Coalition Theory Network organised by the Fondazione Eni Enrico Mattei and the CORE, Université Catholique de Louvain, Venice, Italy, January 11-12, 2002

(lv) This paper was presented at the First Workshop of the Concerted Action on Tradable Emission Permits (CATEP) organised by the Fondazione Eni Enrico Mattei, Venice, Italy, December 3-4, 2001 (lvi) This paper was presented at the ESF EURESCO Conference on Environmental Policy in a Global Economy "The International Dimension of Environmental Policy", organised with the collaboration of the Fondazione Eni Enrico Mattei , Acquafredda di Maratea, October 6-11, 2001

(lvii) This paper was presented at the First Workshop of "CFEWE – Carbon Flows between Eastern and Western Europe", organised by the Fondazione Eni Enrico Mattei and Zentrum fur Europaische Integrationsforschung (ZEI), Milan, July 5-6, 2001

(lviii) This paper was presented at the Workshop on "Game Practice and the Environment", jointly organised by Università del Piemonte Orientale and Fondazione Eni Enrico Mattei, Alessandria, April 12-13, 2002

(Ivix) This paper was presented at the ENGIME Workshop on "Mapping Diversity", Leuven, May 16-17, 2002

(lvx) This paper was presented at the EuroConference on "Auctions and Market Design: Theory, Evidence and Applications", organised by the Fondazione Eni Enrico Mattei, Milan, September 26-28, 2002

	2002 SERIES
CLIM	Climate Change Modelling and Policy (Editor: Marzio Galeotti)
VOL	Voluntary and International Agreements (Editor: Carlo Carraro)
SUST	Sustainability Indicators and Environmental Valuation (Editor: Carlo Carraro)
NRM	Natural Resources Management (Editor: Carlo Giupponi)
KNOW	Knowledge, Technology, Human Capital (Editor: Dino Pinelli)
MGMT	Corporate Sustainable Management (Editor: Andrea Marsanich)
PRIV	Privatisation, Regulation, Antitrust (Editor: Bernardo Bortolotti)
ETA	Economic Theory and Applications (Editor: Carlo Carraro)

2003	SERIES
------	--------

CLIM	Climate Change Modelling and Policy (Editor: Marzio Galeotti)
GG	Global Governance (Editor: Carlo Carraro)
SIEV	Sustainability Indicators and Environmental Valuation (Editor: Anna Alberini)
NRM	Natural Resources Management (Editor: Carlo Giupponi)
KNOW	Knowledge, Technology, Human Capital (Editor: Gianmarco Ottaviano)
IEM	International Energy Markets (Editor: Anil Markandya)
CSRM	Corporate Social Responsibility and Management (Editor: Sabina Ratti)
PRIV	Privatisation, Regulation, Antitrust (Editor: Bernardo Bortolotti)
ETA	Economic Theory and Applications (Editor: Carlo Carraro)
CTN	Coalition Theory Network